Journal Pre-proof

Systematic Review of Internet of Medical Things for Cardiovascular Disease Prevention among Australian First Nations

Khondker Mohammad Zobair, Luke Houghton, Dian Tjondronegoro, Louis Sanzogni, Md Zahidul Islam et al.



PII:	S2405-8440(23)09628-7
DOI:	https://doi.org/10.1016/j.heliyon.2023.e22420
Reference:	HLY e22420
To appear in:	Heliyon
Received date:	22 October 2022
Revised date:	29 October 2023
Accepted date:	12 November 2023

Please cite this article as: K.M. Zobair, L. Houghton, D. Tjondronegoro et al., Systematic Review of Internet of Medical Things for Cardiovascular Disease Prevention among Australian First Nations, *Heliyon*, 9, e22420, doi: https://doi.org/10.1016/j.heliyon.2023.e22420.

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2023 Published by Elsevier.

Systematic Review of Internet of Medical Things for Cardiovascular Disease Prevention among Australian First Nations

Khondker Mohammad Zobair^a, Luke Houghton^a, Dian Tjondronegoro^a, Louis Sanzogni^a, Md Zahidul Islam^b, Tapan Sarker^c, Md Jahirul Islam^d

 ^aDepartment of Business Strategy and Innovation, Griffith Business School, Griffith University, Nathan, QLD, 4100, Australia
^bComputer Science and Engineering Discipline, Khulna University, Khulna, 9208,

Bangladesh

^cUniversity of Southern Queensland, Brisbane, QLD, 4300, Australia

^dGriffith Criminology Institute, Griffith University, Mt Gravatt, QLD, 4122,

Australia

Abstract

Chronic diseases within Indigenous communities constitute the most compelling ill-health burdens and treatment inequalities, particularly in rural and remote Australia. In response to these vital issues, a systematic literature review of the adoption of wearable, Artificial Intelligence-driven, electrocardiogram sensors, in a telehealth Internet of Medical Things (IoMT) context was conducted to scale up rural Indigenous health. To this end, four preselected scientific databases were chosen for data extraction to align with the Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA) technique. From the initially collected (n = 4436) articles, a total of 32 articles were analysed, being synthesised from the review inclusion criteria, maintaining strict eligibility and eliminating duplicates. None of the various studies found on this innovative healthcare intervention has given a comprehensive picture of how this could be an effective method of care dedi-

Email addresses: k.zobair@griffith.edu.au (Khondker Mohammad Zobair), l.houghton@griffith.edu.au (Luke Houghton), d.tjondronegoro@griffith.edu.au

Preprint submitted to Elsevier

^{*}Corresponding author

⁽Dian Tjondronegoro), l.sanzogni@griffith.edu.au (Louis Sanzogni), zahid@cse.ku.ac.bd (Md Zahidul Islam), Tapan.Sarker@usq.edu.au (Tapan Sarker),

Mdjahirul.bd@gmail.com (Md Jahirul Islam)

cated to rural Indigenous communities with cardiovascular diseases (CVDs). Herein, we presented the unique concepts of IoMT-driven wearable biosensors tailored for rural indigenous cardiac patients, their clinical implications, and cardiovascular disease management within the telehealth domain. This work contributes to understanding the adoption of wearable IoMT sensordriven telehealth model, highlighting the need for real-time data from First Nations patients in rural and remote areas for CVD prevention. Pertinent implications, research impacts, limitations and future research directions are endorsed, securing long-term Wearable IoMT sensor-driven telehealth sustainability.

Keywords: Artificial Intelligence, Cardiovascular Disease, Indigenous population, Internet of Medical Things, Internet of Things, Machine Learning, Telehealth, Wearable ECG sensors

1. Introduction

Indigenous healthcare provision is in dire need of delivery reform. A recent study indicates that more than 370 million Indigenous inhabitants live worldwide with low health standards compared with benchmark populations [1]. Indigenous communities bear a heavy burden of illness, leading to lower life expectancy, severe infectious diseases, malnutrition, depression, infant and child mortality, high maternal morbidity and mortality, rising levels of cardiovascular diseases, and other chronic metabolic disease loads [2, 3]. Chronic diseases, including obesity, hypertension, CVDs, Diabetes mellitus, chronic kidney disease (CKD), and renal failure, have become significant health complications worldwide that cause millions of deaths every year [4].

CVD, an umbrella term for heart and blood vessel conditions, is a significant concern in Australian rural indigenous communities [5]. Porykali et al. [6] reveal that the indigenous populations of Australia experience higher rates of CVD, leading to increased hospitalisations and mortality compared to their non-indigenous counterparts. This pattern is not unique to Australia, as indigenous communities worldwide face elevated CVD risks [7]. While chronic diseases associated with health disparities among these communities are well documented, it remains unclear whether existing interventions sufficiently address these issues [8]. Health inequalities persist, with the indigenous population bearing a disproportionate disease burden [9]. Crengle et al. [10] demonstrate a high prevalence of clinical diagnoses related to CVD among indigenous populations in Australia and Canada [10], highlighting the substantial health disparity [9].

Telehealth/telemedicine, particularly when integrated with intelligent wearables through the Internet of Things (IoT)/ Internet of Medical Things (IoMT), offers a potential solution to bridge healthcare access gaps [11]. These techniques substantially impact modern healthcare systems by bringing value to health seekers, providing high-quality, cost-effective services, and promoting effective remote care. This study aims to explore how IoMTdriven telehealth can improve health outcomes and reduce health inequalities among rural and remote indigenous communities in Australia, especially in managing CVD through real-time patient monitoring and optimal disease management [12, 13, 14, 15].

While prior studies have investigated chronic diseases among Australian Aboriginal communities [9, 16, 7, 17], there is a gap in understanding how smart telehealth, particularly IoMT-aided CVD care, can benefit these communities comprehensively. This study presents innovative concepts of IoMTdriven wearable biosensors tailored for rural indigenous cardiac patients and their implications for cardiovascular disease management within the domain of telehealth. This study contributes by identifying adoption determinants for IoMT-driven telehealth for regional CVD care, advocating for real-time care, and shedding light on how IoMT technologies can create a novel telehealth model for remote CVD in Australia and similar settings.

This article proceeds as follows. Section 2 begins with the research rationale and conceptual model. Section 3 labels the research methods used. Section 4 outlines the data analysis and results. Section 5 focuses on a discussion of the findings. Section 6 discusses the contribution and managerial implications. Section 7 proposes the research impacts. Section 8 provides limitations and suggestions for future research directions. Section 9 concludes the article and underlines research highlights.

2. Research Rationale

CVD is the second most significant disease burden in Australia [18]. Among the Australian First Nation people, CVD is the leading cause of disease burden and death and one of four chronic conditions that account for 70% of indigenous Australian health gaps [18]. These Australian communities suffer from heavy infectious disease loads, increasing cardiovascular and other chronic diseases, and overall poorer health indicators compared to their non-aboriginal counterparts [3]. Gibson et al. [17] remarked that chronic diseases predominantly contribute to health disparities since the life expectancy gaps peaked at 50% of Aboriginal and non-aboriginal Australians. These findings exhibit a disproportionate burden of ill health and social suffering upon Australian Aboriginal populations [9]. Unexpectedly, much less attention has been given to reducing severe health burdens, social suffering, and health gaps within this minority group and decreasing health disparity between indigenous and non-indigenous populations. Smart wearable IoMT Sensors and AI-driven telehealth would be well suited to provide cost-effective, high-quality, specialised cardiac patient care and minimise the dominating determinants of health disparity.

A plethora of contemporary research is primarily focused on the application of smart telemedicine (i.e., telehealth) for rural and remote patients using IoT technology [19, 20, 21, 22, 23, 24, 25]. These innovative IoTaided healthcare systems are used in clinical and operational situations as part of digitally transformative practices. For example, Sawyer et al. [25] claimed that physicians make complex clinical decisions using medical big data generated by smart wearable sensors/devices. This supports [26], who asserted that big data holds great promise to create analytic models for better disease predictions, prevention, and management. Morgan et al. [23] suggested that the ground-breaking medical sensors embedded with IoMTdriven telehealth services enable patients to receive enhanced treatments and medical advice remotely. Within telemonitoring, Koya et al. [22] confirmed that an algorithm is designed to run at the gateway node to optimise the power efficiency of the sensor without causing a power drain at the gateway node. Medin-Eastwood et al. [27] stressed that wearable IoT sensors act as enablers, incessantly producing a large volume of information from structured and unstructured medical big data. This validates an underlying contribution made by these traditional technologies and Medical technologies. Broadly these are trending towards an IoT-driven healthcare ecosystem in mainstream healthcare provision.

The IoT refers to the interconnected network of physical objects (i.e., "Things") that are integrated into the exchange of data between devices and sensors through the Internet [28]. Bajao et al. [29] revealed that IoT encompasses a network connected to the internet with various sensors, electrical chips, and relevant hardware components. The combinations of sensors technologies (MedTech) and IoT technologies have the novelty [30] to make connections between people and objects via wearables a convenient to



Figure 1: Conceptual Model of Wearable IoMT ECG AI-driven telehealth.

provide them with a convenient living environment [31]. In another study, Dahlqvist et al. [30] landmark applications in diverse sectors, including smart cities, smart homes, connected cars, and e-Health/telehealth/m-health. In an application of IoT, Dwivedi et al. [28] noted that Wireless Body Area Network (WBAN) systems play significant roles in building IoT-aided Telehealth frameworks for real-time rural patient monitoring, treatment and disease management. Similarly, Riley et al. [24] illustrate how wearable biomedical IoMT sensors and AI-driven telehealth can screen patient physiological complexities and predict severe medical conditions. Dwivedi et al. [28] further pointed out that the IoMT-driven robotic technology can interact with patients after analysing medical data, providing vital signs and the status of their body to predict the risk of CVD, and recommending the necessary lifestyle changes to avoid associated complications.

As mentioned above, the fast-growing innovation of machine learning (ML) and AI, further fuelled the digital transformation in healthcare services to deliver a better patient experience and optimal care. Digital health transformation technologies such as the IoMT, virtual care, real-time re-

Journal Pre-proof

mote monitoring, robotic surgery, AI, Big Data analytics, smart wearables, e-health/telehealth/telemedicine, and m-health platforms are part of modern healthcare services [32]. These technologies enable the storage and sharing of relevant health information across the health ecosystems, improving medical diagnosis, the prognosis of medical risks, creating a continuum of care, and improving health outcomes, thereby creating more evidence-based knowledge for health professionals to support cutting-edge healthcare systems [32]. For instance, IoT mobile, wearable devices and smart medical sensors are instrumental in developing a smart healthcare system (i.e., telehealth). These are omnipresent, fast, and seamlessly accessible to patients [33] living in rural and remote areas. Motivated by prior research on Wearable IoMT ECG AIdriven telehealth [28, 34, 35, 25], the proposed conceptual model identifies the associations between Wearable IoMT ECG determinants and AI-driven telehealth determinants, contributing to developing a future model of care. The conceptual model in Figure 1 investigates the research question.

This systematic review focuses on the clinical applications and evidencebased interventions of wearable IoMT ECG sensors and applications of AIdriven telehealth. In particular, this review focuses on IoT technologies that prevent the risk of cardiovascular diseases and mortality among Australian Aboriginal communities via smart telehealth ecosystems.

Meta-studies [7, 36, 17, 37] suggest research on Australian aboriginal patients with CVD has primarily been focused on conventional health care interventions. Little evidence suggests how wearable AI-driven IoMT electrocardiogram sensors can diagnose and predict Aboriginal patients' risk of CVD regardless of their geographical locations. Moreover, there is scant evidence of a robust IoMT-driven telehealth model of CVD care for these communities. These endeavours are predominantly missing and almost ignored in most existing literature in this critical field. Additionally, there has been little to no investigation into wearable IoMT ECG sensors or AI-driven telehealth models to prevent CVD risks among Australian First Nations living in rural, regional, and remote settings. The determinants of this knowledge gap ought to be filled. The present review examines the following research question:

RQ: How could wearable IoMT Electrocardiogram sensors that use Artificial Intelligence-driven telehealth models be harnessed to prevent cardiovascular disease-causing morbidity and mortality among First Nations peoples living in rural, regional, and remote Australia?

3. Research Methodology

Systematic reviews and meta-analyses are essential tools and techniques for summarising the evidence extracted from the literature precisely, accurately and reliably [38]. This systematic review uses the standard guidelines as the Preferred Reporting Items for Systematic Reviews and Meta-Analysis(PRISMA), an approach recommended by many scholars [39, 38]. PRISMA provides comprehensive review guidelines for each checklist item related to research backgrounds, development, explanations, and rationale that reviewers can follow to ensure the undertaken review processes and meta-analysis are authentic and transparent [38].

3.1. Information Sources

This study conducted a comprehensive peer-reviewed journal article search using four digital databases. The four digital databases, IEEE Xplore, Web of Science, ScienceDirect, and PubMed, provide a broad view of health informatics research from 2015 to 2021 and are deemed appropriate and relevant to the study's discipline. These data sources allowed us to extract a body of scholarly work on the topic, enhancing the credibility of our research findings. In addition, using four reputable databases ensured the inclusion of high-quality articles, further strengthening the validity of our systematic literature review and enabling us to provide valuable insights into the topic and contribute to the existing body of knowledge in health informatics.

3.2. Search approach

The search strategy for the targeted articles was limited to peer-reviewed journals from 2015 to 2021. The initial search was started on 25 July 2020 and was updated on 31 June 2021 for relevant studies that the initial search might have missed [39]. Addressing the PRISMA statement, the following search strings are used for article extraction, "Internet of Things (IoT) on cardiovascular disease prediction" OR "Indigenous Cardiovascular Diseasecausing death prediction through IoT embedded Telehealth", AND "IoTbased telemedicine". We used a single and mix of keywords, including operators "AND" and "OR", such as "Indigenous Cardiovascular disease and Telehealth", OR "Indigenous-CVD-Telehealth", "Australian Telehealth and Indigenous CVD", "Internet of Things and Cardiovascular disease", "Australian Rural Health and Aboriginal people", "Indigenous people and Telehealth", "Cardiovascular Disease and telehealth", AND "CVD and IoT based telehealth, OR IoT and Telehealth", "Aboriginal people and Telehealth", OR "First people and telehealth", "First People and Cardiovascular disease", "Torres Islanders and Telehealth", "Machine learning and Internet of Things based wearables", AND "Deep learning and Cardiovascular disease monitoring using wearable IoMT sensors", OR "Artificial Intelligence and Cardiovascular disease" AND "AI and CVD" OR "IoT and wearable ECG devices", AND "IoT and wearable ECG sensors", OR "Wearable CVD monitoring devices".

3.3. Inclusion, Exclusion and Eligibility

This review included (inclusion criteria) strictly peer-reviewed published journal articles in English related to the mainstream of research; others, including duplications along with irrelevant resources, are excluded (exclusion criteria). Furthermore, this review included studies involving wearable and handheld electrocardiogram sensors, IoT technologies, as well as AI and machine learning applications. However, studies not directly related to cardiovascular disease detection, prevention, and management were excluded from consideration. It is worth noting that this review included peer-reviewed articles related to Aboriginal patients with CVDs as part of the search due to the unavailability of sufficient relevant research on wearable IoT-aided electrocardiogram sensors and AI-driven telehealth for Aboriginal patients with CVD in rural and remote Australia and globally. To ensure the appropriate article selection, the authors administered three rounds of screening and filtering processes and removed all irrelevant articles, book chapters, conference papers, and research notes, securing the most relevant items for analysis and synthesis. By adhering to stringent inclusion and exclusion criteria and employing meticulous screening and filtering processes, this review maintained a high standard of article selection, contributing to the accuracy of the study findings. The following section discusses the major themes of this research and the outcomes of the analysis. This is focused on the development of an analytic model for CVD risk prediction and policy recommendation for sustainable IoT/IoMT aided telehealth interventions for the rural and remotely living Indigenous communities.

4. Analysis

This study presents the review results using PRISMA guidelines (see Figure 2) at each stage of the article classification (i.e., cleaning and filtering),



Figure 2: Literature search and selection process using PRISMA technique



Figure 3: Articles published globally optimising Wearable ECG Sensors for CVD care.

adhering to the selection processes. Continuing with this approach, initially, this research yielded (n = 4436) articles from four digital databases, namely IEEE Xplore, Web of Science, ScienceDirect, and PubMed. Upon the complete screening of the title, abstract, and content and duplication, a total (n = 190) of papers were primarily selected that were deemed relevant to the purpose of this study. The filtering process was administered, and there were (n = 56) articles included for congruency with the review inclusion criteria and (n = 135) papers were excluded. Full-text articles were further assessed to meet eligibility criteria since, after careful appraisal (n = 24), articles were excluded due to not being strictly pertinent to the study's subject matters. Finally, a total of (n = 32) articles were finalised for data synthesis (see Figure 2).

To summarise the existing findings, Figure 2 provides evidence of the literature search and selection process administered using the PRISMA review technique. While Figure 1 demonstrates the conceptual model as the workflow of wearable IoMT ECG sensors, the AI-driven telehealth (IoMT, AI-TH) model for the constant real-time monitoring the patients with CVD in rural and remote areas is relevant to this study. In the present study, Table A.1 (see Appendix A) and Table B.2 (see Appendix B) describe the characteristics



Figure 4: Summary Characteristics of Studies on Wearable IoT-ECG Sensors, AI-Driven Telehealth for CVDs Care.

of the articles published within the time frame. Figures 3 and 4 exhibit the summary characteristics of diverse applications of wearable IoT-aided ECG sensors and AI used to diagnose and prognosis the CVD described in the literature. Moreover, our findings indicate that (19%) of studies validate the compatibility of wearable IoT/IoMT ECG sensors with AI-driven telehealth for CVD care across diverse settings. The key findings to be addressed in the following sections are the reviews.

4.1. Statistical and Geographical Distributions of Published Articles

Figure 5 shows the number of articles published from 2015 to 2021, indicating an overall increasing tendency, particularly in the USA (25%), accounting for eight articles, followed by China (15.62%), and India (12.5%) each contributing five and four respectively. Similarly, Australia (9.37%), France (6.25%) and Poland (6.25%) each produced 3 and 2 articles, respectively. Meanwhile, the UK (3.12%), Germany (3.12%), Belgium (3.12%), Italy (3.12%), Singapore (3.12%), Malaysia (3.12%), Iran (3.12%), and Pakistan (3.12%) each presented similar papers in the review. The inclusion of works from various countries offers a broader view of IoT/IoMT sensorsdriven telehealth models for rural CVD care, ensuring a more holistic analysis and synthesis of the research landscape.

This distribution provides deep insight into the global interest and active engagement of various countries in research related to Wearable IoT-ECG Sensors, AI-Driven Telehealth for CVDs care. This validates that the USA, China, and India have been particularly active in generating new findings



Figure 5: Statistical Distribution of Articles Published in Several Countries.

on this innovative model of care, while other countries also have important contributions with fewer publications.

Furthermore, the varied geographical distribution of research contributions from countries like the USA, China, and India (see Figure 6) reflect their dominant engagements in this emerging field of research, reinforcing the innovative nature of this model of care. The variation in the number of studies from these countries suggests disparities in research emphasis, adequacy of funding and prevalence of the topic interest. This distribution also helps identify potential collaborations in research within these countries, offering significant opportunities for further exploration and cross-country comparisons. Additionally, the insights gained from this diverse pool of literature will better identify the determinants of adopting IoT/IoMT wearable ECG sensors and AI-driven telehealth models within Australia and similar settings. Finally, the findings corroborate the identification of potential gaps and opportunities for furture research and practices within this innovative field.

4.2. Wearable IoT/IoMT Sensors, AI-driven Telehealth Platform

IoT/IoMT-driven telehealth services have gained intensive attention and sparked interest in successful adoption to create clinical and economic val-



Figure 6: Geographical Distribution of Articles Published between 2015 to 2021.

ues amongst diverse global health providers. The continuous evolution of technology and artifacts plays a key role in developing IoT/IoMT-based telehealth services to ensure high-quality service, safe and secured health access with affordable and reliable coordinated care for rural and remote patients. Furthermore, with the growing recognition of the roles of the IoT/IoMT and the multi-purpose biomedical devices and their interoperability of functions, there is an increased impetus for building patient-centric 'smart' healthcare systems. For example, Jin et al. [40] asserted that the IoMT technology has gradually been used in remote patient monitoring, screening and treatment using various innovative (i.e., MedTech) medical sensors and devices. The authors proposed a multi-dimensional predictive model based on BP neural network [40]. The proposed study incorporates multi-dimensional data analysis and achieves high prediction accuracy as an important guiding significance for intelligent medical treatment.

To determine the sustainability of IoMT-based telehealth for cardiovascular care in rural settings, Hamil et al. [21] proposed a wearable IoMT ECG, AI-driven telehealth model for arrhythmias of cardiovascular disease detection, prediction and management. The proposed wearable ECG can capture the bio-signal data and analyse them using AI and ML approaches. The authors found this model has achieved high prediction accuracy peaking at 99.56%, demonstrating the future opportunity of a large-scale deployment. Throughout this review, we will use IoT and IoMT interchangeably [21]. Utilising IoT in the MedTech industry context, Albahri et al. [19] mapped the lifecycle and architecture of wearable IoT-based telemedicine (i.e., telehealth) healthcare framework comprising IoT wearable sensors, network communications, cloud computing, hardware devices, smartphones, and AI technique. The authors in [19] further show that the IoT-based healthcare systems (Lo-RaWAN) contain four essential elements: data collector, analysis, treatment plan generator and treatment plan executor. Similarly, Grooby et al. [41] designed a wearable IoT phonogram (PCG), an AI-driven telehealth model for the automatic estimation of heart rate (HR) and breathing rate (BR) of CVD care. The performance and reliability of signal detection have been tested for the proposed PCG [41]. A total of 88 ten-second-long chest sound samples were taken from 76 preterm and full-term babies since this PCG successfully detected high-quality sound data and analysed them using AI and achieved a high prediction accuracy of heart rate and breath rate accounting for 93% and 82%, respectively, elucidating a robust model of CVD care for neonatal CVD patients for telehealth applications.

Taking a new approach to wearable IoT ECG sensor-based telehealth for CVD management, Koya et al. [22] distinguished the era of the IoT and hyperconnection with an ECG telemonitoring via WBAN or cloud-based within telehealth framework. The authors found these models are easy to use, self-configured, secure, plug-and-play systems with minimum hardware. Furthermore, Koya et al. [22] investigated the adaptability of smartphones as an IoT gateway for sending and receiving data to a remote server. Mobile IoT gateways offer high potential due to their widespread usage, small size with relatively high computational power, and seamless wireless connectivity [22]. Similarly, integrating more technologies into wearable IoT ECG sensordriven telehealth systems, Beach et al. [12] proposed a wearable IoT Wrist-Worn ECG sensor to monitor home-based patients with CVDs in out-ofclinic settings in the UK. This wearable offers low-power consumption for real-time CVD patients' heart rate variability monitoring systems. These studies conclude that wearables IoT/IoMT ECG sensor has great potential for rural and remote CVD care.

Within IoMT-enabled applications, Sanamdikar et al. [15] proposed an IoMT-integrated electrocardiogram (ECG) sensor (IoMT) to monitor five different forms of beat arrhythmias, including regular, supraventricular ectopic beats, ventricular ectopic beats, the fusion of ventricular and normal, and fusion of placed and normal (N, S, V, F, Q) for early detection of heart problems of CVD patients. The findings reveal that the proposed algorithm successfully predicted cardiac arrhythmias, accounting for 98% of accuracy in feature extraction, classifications, and arrhythmia detection validating tremendous value to the patients with CVDs [15]. Additionally, the authors further clarified how the ECG beats classification technique can improve accuracy, sensitivity, specificity, and precision associated with detecting cardiac arrhythmias [15]. Dwivedi et al. [28] presented a comprehensive guideline of the IoMT structure and its competitive advantages in the healthcare system, along with various potential applications. The authors classified components of IoMT into several categories, given the full details concerning the methods and analysis used in the study.

From the prolific growth of MedTech, Rashid et al. [42] proposed a portable ECG-based telemedicine model for real-time monitoring of patients with CVD residing in non-clinical environments such as the home, office, or remote rural areas. This lightweight, portable ECG sensor enables sensing patients' heartbeat, amplitude level, and PQRST wave via the Atmega-32 microcontroller using the RS-232 serial module. Likewise, Randazzo et al. [43] designed wearable, wireless-based ECG Watch wrist-worn sensors for real-time patients with CVD monitoring. The proposed algorithm detects possible atrial fibrillation episodes within 10 seconds through a smartphone or desktop App. The authors verified that this wrist-worn ECG Watch sensor performed well in diagnosing and prognosis of atrial fibrillation disease since this cannot be easily detected in reality [43]. Interestingly, both proposed (portable ECG and wrist-worn ECG sensors) are designed for telehealth applications, but none was implanted with IoT/IoMT platforms. Likewise, Wang et al. [44] proposed a dynamic ECG compatible with telemedicine to prevent and diagnose CVD patients in Singapore. The authors applied ECG signal analysing algorithms for external noise reduction captured by ECG and found noise reduction performance outstanding. However, these sensors could have been tested with IoT/IoMT platforms for validity checks within the telehealth domain. To identify what elements of wearable IoT/IoMTbased ECG sensors are compatible with the telehealth framework for rural and remote CVD care.

4.3. CVD Challenges in Australian Aboriginal Communities

Research indicates that chronic disease is the single leading cause of death among Aboriginal and Torres Strait Islander peoples in Australia [17]. Gi bson et al. [17] reviewed and highlighted the enablers and impediments related to adopting a primary healthcare model to support indigenous populations with chronic diseases in Australia, New Zealand, Canada and the USA. Their findings have some similarities to the other publications. For instance, Calabria et al. [7] found that the CVD risk is consistently higher in indigenous than non-indigenous populations indicating that Native American, Canadian First Nation, Māori, and Australian Aboriginal and Torres Strait Islander communities are at increased risk against their counterparts. Interestingly, the authors found a high absolute CVD risk in young Australian Aboriginal and Torres Strait Islanders under the age of 35 years [7], vividly highlighting health inequalities between the two groups. Unfortunately, research, strategies, policy implications, and individual and community-based efforts are not being sufficiently put forward to recognise the realities of chronic diseases. Health and well-being are also being ignored, pointing to direct and indirect health disparities between Aboriginal and non-aboriginal communities [9].

Apart from other chronic diseases, coronary heart disease (CHD), which may appear without any symptoms of cardiovascular disease, is the leading cause of morbidity and mortality worldwide [45]. The authors presented an invasive CVD diagnostic measurement and evaluation of aortic stiffness in the carotid-femoral pulse wave velocity (PWV) index to diagnose patients with CVD [45]. This index explains how the velocity of arterial pulse moving along the vessel wall indicates and predicts possible CVD events. The authors further clarified the antecedents to CVDs causing events comprising age, sex, blood pressure (BP), and heart rate (HR), all tied to be substantial auxiliary factors of aortic stiffness representing an essential index for the CVD diagnosis [45]. Moreover, they used cutting-edge AI (i.e., Artificial Neural Network) technology to explore a new CVD characteristics/elements pattern that could effectively detect coronary heart disease, prevention, and management [45].

From an early detection point of view, Shomaji et al. [46] proposed a novel wearable diagnostics system for CVD patients in the USA. These authors outlined a set of CVD diagnostic tools that can assist physicians in detecting heart diseases, including CT heart scan, chest X-rays, blood tests, cardiac catheterisation, heart MRI, stress test, pericardiocentesis, myocardial biopsies, and coronary angiography. Further, they designed the essential hardware components for wearable imaging systems and an algorithm to predict intima-media thickness (IMT), an indicator of CVDs. Likewise, in Belgium, De et al. [47] offered a multi-parameter wearable sensor for follow-up cardiac rehabilitation patients. The authors applied AI and ML approaches to visualise the relationships between sensor-derived biomarkers and sensors' capability to monitor remote CVD patient tracking. Lin et al. [35] presented an IoT-aided wearable ECG sensor for real-time patient monitoring with CVDs. They found the five different sensors that can be used to detect cardiovascular diseases such as Pulse Wave Velocity (PWV), electrocardiogram (ECG), phonocardiogram (PCG), Seismocardiogram/ballistocardiogram (SCG/BCG) and apexcardiogram (ACG). The same was found in the study by Yang et al. [48], who proposed an IoT-cloudbased ECG sensor for real-time patients with CVDs monitoring. These authors classified bio-signal data into five different ECG signal categories (P wave, T wave, Q wave, R wave and S wave) that should allow physicians to diagnose cardiovascular diseases [48].

Creating value in chronic care for Aboriginal populations via telehealth, Brazionis et al. [16] proposed a telehealth model for remote and very remote indigenous patients with CVD and Diabetes. Due to insufficient data availability, the authors could not strongly conclude whether telehealth applications can facilitate best practices in CVD and Diabetes care and disease management in remote indigenous communities in Australia. This unanswered question demands an avenue of research for this exciting new area. Table A.1 and Table B.2 (appendix) provide evidence that almost all studies used wearable ECG sensors for CVD care. At the same time, the aggregation of findings Tables A.1 and B.2 (appendix), and Figures 3-6 confirm that existing literature recognized the potential significance of IoT/IoMT-ECG sensors AI-driven telehealth for rural and remote CVD care.

4.4. Wearable IoT/IoMT ECG Sensors-based CVD Patients Monitoring

In Figure 3 and Table B.2 (appendix), diverse applications of wearable IoT-aided ECG sensors to diagnose and prognosis CVD reported in the literature were exhibited. Much work on various wearable, handheld, and dynamic ECG sensors integrated into IoT, artificial intelligence-based algorithmsdriven sophisticated systems to real-time monitoring patients with CVDs proposed by Hamil et al. [21]. For instance, various ECGs are made of flexible materials, lightweight, and low-cost and are successfully and purposefully used for biomedical sensing. For example, Balsam et al. [49] proposed a biomedical shirt-based electrocardiography (ECG) sensor to monitor patients with CVD in various clinical situations. The proposed wearable ECG sensor uses Nuubo ECG systems, enables monitoring of patients with CVD, captures high-quality ECG recordings and ensures comfort for patients while wearing a biomedical shirt [49]. This novel wearable ECG technology was tested using four independent patient groups with CVD comprising patients after pulmonary veins isolation (PVI) procedure, cardiac resynchronisation therapy recipients(CRT), patients during cardiac rehabilitation after the myocardial infarction, and paediatric patients with supraventricular tachycardia (SVT) [49]. The authors found this highly effective in improving CVD diagnosis in different situations since this is washable, allowing greater patient comfort and cost-effectiveness. The biomedical shirt ECG is used in continuous real-time recordings with a battery life lasting up to 36 hours [49]. Additionally, this wearable ECG has become viable for clinical applications certified by European Union [49].

From advances in medical sensors and leveraging patients' care point of view, Lin et al. [35] summarised various sensor technologies and their flexible bio-signal sensing mechanisms, what is known as an electrocardiogram (ECG), phonocardiogram (PCG), seismocardiogram/ballistocardiogram (SCG/BCG), and apexcardiogram (ACG) used for managing cardiovascular diseases in China. Lin et al. [35] explicated how these sensors capture bio-signals, pulse wave signals, and the characteristics/elements that play key roles in CVD incidents. Similarly, Borujeni et al. [50] presented a four-layer IoT-driven intelligent healthcare system for real-time monitoring of patients with cardiovascular diseases. In the proposed model, a patient's vital signs are measured using a body sensor network and sent to an intelligent healthcare domain. Their findings confirm a significant improvement of 70% in response time and scalability compared to the state-of-the-art techniques. Another study by Al-Alusi et al. [51] configured several groups of wearable ECG sensors comprising AlivCor Kardia devices, AlivCor Apple Watch Series 4, and several others are available for clinical use since these wearables are commonly integrated into the health network infrastructure. For instance, QardioCore and Hexoskin are chest-worn-based sensors capable of recording high-quality ECG signals and tracking patients when placed in their bodies [51], thereby helping physicians manage patients with CVDs even remotely.

By optimising CVD care, advancing detection and diagnosis via ECG sensors, Baghel et al. [52] designed a phonocardiogram (PCG) for automated and real-time cardiac disease diagnostic systems that detect the present heart conditions of remote patients. Baghel et al. [52] used machine learning, AIbased (i.e., convolutional neural network) algorithms for biosignal data classifications and analysis captured by PCG sensors. They achieved high model accuracy, peaking at 98.60%, validating the robustness for multi-cardiac disease prevention, prediction and management [52]. The authors further highlighted the functional features of the PCG, which is compatible with any computing device, single-board computing processors, and Android handheld devices. Similarly, McRae et al. [53] proposed a Cardiac ScoreCard a multivariate index assay system for early detection and frequent monitoring of traditional risk factors along with novel biomarkers for patients with CVDs. The proposed Cardiac ScoreCard system exhibited high-performance functionality and diagnostic accuracy. Another study by Pevnick et al. [54] presented broadened features of existing medical wearables adopted and accepted by many physicians for patients' heart rate and heart rhythm thoracic fluid monitoring. Behind these, to better understand characteristics and reap the benefits of various other wearables points of view, authors provided recommendations for future wearables and their potential in disease management and adoption impediments that must thoroughly be addressed [54].

With the pervasive application of wearables (i.e., ECG) from clinical practices, Sanamdikar et al. [15] implemented an IoT-based ECG that categorises five different beat arrhythmias (N, S, V, F, U), which are essential to identify a patient's heart problems. The evidence from their findings suggests that the device remained pivotal in screening, detection and prediction of cardiac arrhythmias than other approaches and achieved high predictive accuracy accounting for 98% [15]. This supports Beach et al. [12] findings, who proposed a wearable IoT Wrist-Worn ECG sensor to monitor homebased patients with CVDs in out-of-clinic settings in the UK. The authors confirmed that the proposed wearable IoT Wrist-Worn ECG sensor is significantly effective, user-friendly and lightweight 50g, including the strap, has low power consumption, and is compatible with SPHERE (Sensor Platform for Healthcare in a Residential Environment) smart home architecture [12]. Similarly, Florez et al. [55] demonstrated an efficient wearable, IoT-aided BlooXY sensor for cardiovascular disease control, prevention, treatment, and management. The proposed IoT-based BlooXY sensor can sense and monitor patients' blood pressure, heart rate, and blood oxygen level (oximetry-SPO2), which are the essential characteristics of CVD detection and prevention [55].

4.5. Applications of AI and ML in CVD Care

Widespread adoption interest in machine learning technology has revitalised the field of data science, and AI-powered applications have become a driving engine in many organisations [56], including health care. Tsay et al. [56] provided an overview of how AI and ML approaches promote realtime care for patients with CVDs. In this context, the authors outlined the strategies to strengthen existing clinical processes to increase accessibility, effectiveness, efficiency and availability of CVD care. Bini et al. [57] asserted that ML as a subset of AI is experiencing exponential growth in healthcare applications and has a profound impact on care delivery refinement. The author's purpose is to demystify these technology innovations for practising data scientists so they can better grasp how and where to apply them [57].

AI applications for diagnosis and prognosis are sustained in various branches of health, including oncology, dermatology, neurology, and cardiology [21]. For instance, Raj et al. [14] proposed automated handheld arrhythmias detection ECG systems for CVD diagnosis in India. The authors presented high-performance metrics that yield an overall accuracy peaked at 92.81%, 92.68% and 92.42% with average sensitivity, specificity and positive predictivity, respectively. Krittanawong et al. [58] asserted that deep learning (DL) is well-suited to cardiovascular medicine. Figure 4 illustrates that the wearable IoT/IoMT ECG sensors AI-driven telehealth for CVD research peaked at 38% between 2015 and 2021. Likewise, ML, DL, and AI-powered diagnostic tools used for various wearable ECG sensors' sensing bio-signal data analyses for CVD detection, prediction, management and control articles peaked at 66% since different algorithms used papers peaked at 84%. In addition, wearable IoT/IoMT aided ECG, AI-driven telehealth/telemedicine/e-Health/Smart Health papers peaked at 53%. These findings (see Figure 4) validate the continued growth of digital healthcare infrastructure along with virtual CVD care research in this exciting field.

A plethora of literature indicates that cardiovascular diseases are largely preventable but unpredictable due to underlying risk factors that may appear without any symptoms or compliance [45]. Taking this severe challenge of developing novel ML and AI-driven methods for CVD risk prediction is of immediate scientific and practical interest [45].

Similarly, Faust et al. [59] proposed a cost-effective hybrid IoT and advanced AI-based Heart Health Monitoring Service Platform (HHMSP) for CVD management. The proposed hybrid model advocates that humans and computers work together to improve cost efficiency while maintaining the reliability of diagnosis and prognosis of the CVDs [59]. From ML and AI in clinical care point of view, Bini et al. [57] presented how AI can act as a tool amplifying human cognitive functions for health providers to provide healthcare support to increasingly complicated patients. Krittanawong et al. [58] revealed that deep learning is deemed an appropriate method for cardiovascular medicine. Hemodynamic and electrophysiological indices are constantly captured by wearable sensors and image segmentation in cardiac imaging. Tsay et al. [56] demonstrated how AI platforms improve the operational delivery of cardiac care. These corroborate that AI and ML integrated into the health domain should keep pushing forward towards the novel future journey of chronic care for rural and remote communities.

From an IoT, phonography AI-driven telehealth point of view, Grooby et al. [41] offers a new approach to heart rate and breathing rate estimation from noisy neonatal chest sounds. The evidence from their proposed model demonstrated high accuracy in prediction, accounting for 93% heart sound and 82% lung sound, bolstering future telehealth applications for CVD detection, prevention, treatment, management and control in rural and remote Australia. Another study by Hamil et al. [21] designed a secured IoT, ECG AI-driven telehealth for predicting the automatic identification of arrhythmias (cardiac state) and achieved high accuracy peaking at 99.56%. The proposed model showed functional robustness, allowing a good balance between low costs and high performance while maintaining ease of use with prompt access to multiple bio-signals, thereby preventing loss of life during patients' critical situations [21].

Ma et al. [60] investigated Atrial fibrillation (AF) for CVD events. They found it is one of the most common arrhythmias related to CVDs which is difficult to monitor in real-time monitoring due to its intermittent nature. These authors proposed a wearable ECG, AI-driven telemedicine for AF detection and prevention. The proposed model achieved the highest sensitivity accounting for 99.3%, specificity of 97.4% and prediction accuracy of 98.3%, demonstrating an outperforming model of CVD care [60]. Similarly, Wang et al. [44] presented how a dynamic ECG sensor embedded with telemedicine can be implemented for real-time patient monitoring and diagnosing and preventing CVD events. The authors used a deep neural network to demonstrate how external environmental interferences (noises) could be reduced from the dynamic ECG signal classifications [44].

From an AI-powered ECG signal analysis and prediction point of view, Al-Alusi et al. [51] revealed that sensor manufacturers create algorithms that interpret ECG sensing bio-signal data governed by all the same parameters, such as negative predictive values. For example, Apple Heart Study sets a target sensitivity and specificity for their devices (Apple Watch Series 4), accounting for 92% and 90%, respectively, for AF detection algorithm [51]. A recent study by Baghel et al. [52] presented the performance and prediction accuracy of various algorithms used for analysing phonocardiogram (PCG) signal data to diagnose and prognosis of cardiac diseases. The proposed algorithms of Support Vector Machine (SVM), Random Forest (RF), Artificial Neural Network (ANN), Deep Neural Network (DNN), K-Nearest Neighbour (K-NN), Convolutional Neural Network (CNN) without augmentation, and Convolutional Neural Network (CNN) with augmentation peaked at 87.65%, 97.78%, 95%, 89.30%, 96.50%, 96.23% and 98.60% respectively leading its high accuracy and robustness to automatically diagnose and predict cardiac disorders from the PCG signals [52]. Similarly, Hamil et al. [21] presented a novel wearable IoT, ECG sensor, AI-driven telehealth model with secure wireless transmission and classification of the bio-signal platform and Xbee module with Arduino Uno and Raspberry Pi as data acquisition and processing. Authors used ECG signal data for arrhythmias (i.e., CVD event) prediction using different AI algorithms and ML methods, comprising ANN, CNN, SVM, KNN, and RF and the best classification accuracy achieved accounting for 99.56%, [21].

5. Discussion of the results

This is the first study to shed light on the feasibility of adopting wearable IoMT ECG sensors, an AI-driven telehealth model for rural communities, especially suitable for Aboriginal patients with CVDs living in rural and remote Australia. This review revealed that the wearable IoMT ECG sensors' AI-driven telehealth model (see Figure 1) delivers tremendous value to rural patients in an innovative way transforming their journey towards preventive and predictive CVD care. Figure 3 summarises IoT/IoMT embedded wearable ECG, AI/ML, algorithm, and telehealth as analysed from the reviewed studies. The aggregation of findings from the analysis (see Tables A.1 and B.2 (Appendix A and Appendix B), and Figures 3, 4, 5, and 6) concurs that the adoption of the wearable IoMT ECG sensors AI-driven telehealth continues to accelerate potential opportunities in reducing health inequalities between urban and rural counterparts. The results from the studies from 2015 to 2021 (see Figure 3) have grown exponentially and exceeded the

numbers from previous years concerning IoT/IoMT-wearable ECG sensors and AI-driven telehealth for CVD care. These findings validate the assertion that empirical research strongly emphasises the significance of this innovative model of care. This could unfold future models of CVD care for rural, regional, remote, and very remote patients with CVD regardless of the indigenous and non-indigenous communities in Australia and similar settings. Referring to health disparity between two groups, Power et al. [61] found stark health disparities between Aboriginal and non-aboriginal Australians. Another study by Haynes et al. [1] described the colonial legacies resulting in trauma, loss, and grief, contributing to a range of inequitable health and well-being outcomes. Prior research by Adelson et al. [9] asserted that health disparities point to underlying various causes of the imbalances that constantly reside outside the typically constituted health domain.

Empirical evidence from the review shows that chronic disease threat differs between the indigenous and non-indigenous populations in Australia and globally. The assumptions from the study provide evidence that CVDs have a potential impact on human health in general since the treatment period is extended, thus posing a significant threat to patients' health [35]. On this basis, a recent study by Heraganahally et al. [62] demonstrate chronic respiratory conditions among indigenous inhabitants are highly predominant, particularly in English-speaking countries. However, there appears to be significant knowledge gaps concerning indigenous inhabitants in non-English speaking countries. Haynes et al. [1] show how chronic Rheumatic heart disease predominantly impacts young people with the contemporary age-standardised occurrence at 60 times higher in the Australian aboriginal population than non-Aboriginal Australians < 55 years of age. We found that demographic characteristics are dominant factors contributing to chronic disease prevalence between indigenous and non-indigenous groups. For example, Brown et al. [63] revealed that the Indigenous population's age-adjusted cardiovascular disease death remained the most significant single cause of death and was three times higher than in the non-Aboriginal community.

With regard to age factors dominantly influencing high mortality, Brown et al. [63] further illustrated that age-specific CVD causes mortality rates to even worsen between the ages of 25 and 54, peaking at 7 and 12 times that of non-Indigenous populations. Geographical factors also significantly impact chronic disease conditions among indigenous communities. For example, CVD-causing morbidity and mortality ratios also provide important insight into the cardiovascular disease burden for rural and remote aboriginal inhabitants in Australia. In covering these issues, a recent study by Gaffney et al. [64] asserted that rural and remote residents have inadequate resources to treat and prevent chronic obstructive pulmonary disease (COPD) than their urban counterparts in America. This validates that facing higher costs involving chronic conditions, which are more challenging and expensive to treat, patients are less likely to visit physicians due to additional expenses [64]. Our findings suggest that these statistics could consider a more holistic approach to adopting wearable IoMT ECG and AI-driven telehealth systems to tackle chronic disease-causing morbidly and mortality risks and reduce health inequalities in these underserved communities.

The findings from the review confirmed that tackling a growing number of patients with cardiovascular, pulmonary, and metabolic chronic diseases requires a closer look at their symptoms [65]. This corroborates that managing these diseases remains a complex clinical task because it occurs with comorbid conditions. Effective medical treatment of these chronic diseases typically requires lifestyle and food habit changes, medication regimens, and close patient monitoring [66]. Research suggests telehealth is remarkably consistent with satisfying patient care requirements in a challenging healthcare environment [25]. To this proposition, Butten et al. [67] argue that telehealth is persistently valuable and relevant to provide primary and specialist health care for disadvantaged communities who often have unfavourable health access to mainstream healthcare compared with the general population. From a practice point of view, we found that 19% of studies confirm that wearable IoT/IoMT ECG sensors are compatible with AI-driven telehealth for CVD care. The remaining 81% of studies have not been explored within the telehealth domain, indicating a significant gap in the research that our conceptual model aims to bridge.

The evidence from the review shows that the IoMT remained a fascinating digital innovation that seems poised to cross over into human biology, technology, and medical devices to treat rural and remote communities with specialised primary care via telehealth ecosystems. For instance, Albalawi et al. [20] revealed that the IoMT enables interconnecting patients, health providers, medical devices, and machines to promote evidence-based, safe, secure, and reliable patient care. This is consistent with Zhu et al. [68], who highlighted that telehealth is shown to be tied to the effectiveness of reducing risks of heart failure, diabetes, and other chronic diseases, maintaining successful distance communications between patients and physicians and increasing patients' health outcomes compared with conventional healthcare systems. This suggests wearable AI-driven telehealth provides potential health solutions for rural and remote CVD care. This is congruent with Cronin et al. [69] explanation of smart healthcare monitoring systems using cardiovascular devices (CIED: cardiac implantable electronic devices) for remotely living patients conferred a 50% relative decrease in CVD causing deaths than attending clinics follow-up. Similarly, Yang et al. [48] offers a portable ECG integrated into an IoT-based monitoring system to diagnose remotely living patients with cardiovascular diseases. From a low-cost wearable ECG sensor on the mobile devices context, Martinez et al. [70] proposed a wearable ECG sensor from e-Health (i.e., Telehealth) Biometric Sensor Platform designed by Libelium could be used for real-time CVD patients' heart rate variability monitoring.

The findings suggest that electrocardiogram (ECG), blood pressure (BP), and blood oxygen saturation level (SpO_2) sensors integrated into IoT-driven telemedicine enable the collecting of data from remote patients with chronic diseases (i.e., CVD). Secondly, transmitting results can be close to real-time through a remote server connected with computers located in the medical centre [71]. For instance, Albalawi et al. [20] asserted that patients with chronic diseases get alerts if their health conditions deteriorate. The sensors instantly send the recorded information to the physicians via digital health networks. As the above review demonstrates, there is growing support for implementing this technology. This is consistent with our review as it outlines various dynamic electrocardiograms for real-time cardiovascular patient monitoring and demonstrates the tremendous growth of these MedTech devices in recent years. For example, Beach et al. [12] presented a wearable wristband ECG integrated into an IoT-driven model for CVD care. Similarly, Scheffler et al. [72] designed wearable wristband ECG sensors that are compatible with telemedicine and suitable for rural patients with CVD care. These validate that this wrist-worn ECG is ideal for detecting, predicting and managing CVD care, especially for the populations (i.e., patients) living in rural regional and remote areas. For example, Cugliari et al. [73] employed machine learning and AI approaches to predict the biomarkers (i.e., biological molecules found in blood, body fluids, or tissues are the sign of disease conditions) of CVD in Italy and achieved high prediction accuracy, peaked at 90%. These authors describe how myocardial infarction, acute coronary syndrome, ischemic cardiomyopathy, coronary (carotid) revascularisation, and ischemic or haemorrhagic stroke play a key role in CVD events.

In experimental research, Pevnick et al. [54] revealed that wrist-worn

ECG could measure heart rates with less than 10% error compared to slandered devices under ideal circumstances. However, these devices remain largely outside usual channels [74]. More specifically, these devices detect, transmit, store and analyse data but in a database not linked to and incompatible with traditional health records resulting in useful information being unavailable for the physician unless patients volunteer it [74]. This suggests today's healthcare systems have yet to take full advantage of IoT/IoMTenabled sensors/devices to provide extensive medical support and keep patients healthier longer [74]. However, Kindle et al. [75] argue that integrating decision support systems (CDSS) to real-time remote patient monitoring by physicians remained a formidable challenge. It is important to recognise that the advancement of ML algorithms and large databases for CDSS development provide substantial hope that a renaissance in tele-ICU care (intensive care unit) is coming soon [75]. Likewise, Liu et al. [76] proposed classification and recognition methods of encrypted ECG data based on neural networks and found satisfactory accuracy, efficiency and feasibility compared to other solutions.

From the characteristics, adaptability, and compatibility of wearable ECG sensors, this review revealed interest in whether these novel MedTech sensors can benefit patients with CVDs. For example, Al-Alusi et al. [51] asserted that the ECG sensor technology is currently being built into wearable forms capable of real-time monitoring, diagnosing and prognosis of remote patients with CVDs. Similarly, Dwivedi et al. [28] revealed that wearable medical devices with in-built sensors enable the screening of various human body infections and transfer data to monitor the real-time status of symptomatic patients. Majumder et al. [77] confirmed that an electrocardiogram (ECG) is a non-invasive approach commonly used by physicians for measuring the different forms of arrhythmia diseases (i.e., CVD events). Although many arrhythmias are uncategorised as life-threatening, such as myocardial infarction (MI), it may lead to cardiac arrest if not responded to immediately [77]. The review demonstrated various flexible MedTech sensors are currently being used in recent years. Compared with hospital devices, wearables are smaller, have lower power consumption, and can be worn comfortably [78]. These include wristbands, smart watches, glasses, body metric textiles, and more [78]. The advantages of scalability, flexibility, lightweight, and cost-effectivity, polymer films or fabrics advocate designing diverse wearable biomedical flexible sensors [79].

5.1. Recent Advancement (2022-2023)

With recent advancements in current literature, further findings have emerged to strongly support our case. For instance, Blake et al. [80] studied Cardiac Analytics and Innovation for CVD care from an Australian perspective and highlighted the 'siloed' and poorly linked nature of its healthcare data. In contrast, Deniz et al. [81] found that the determinants of AI and Big Data in m-Health adoption in remote care facilities were impossible without ensuring data privacy, security, and quality assessment. IoT-driven eHealth research by Sun et al. [82] found that the high classification accuracy of interpatient ECGs is crucial in diagnosing Arrhythmia (CVD), given the pertinent risk of misclassification in eHealth settings. To resolve such problems, DL methods must be implemented to maintain high classification accuracy in AI-driven eHealth for CVD care [83].

Similarly, to improve the accuracy of IoT-driven ECG sensors, data fusion algorithms that outperform the baseline "20 Channel RR-Interval" averaging approach by $\simeq 54\%$ and $\simeq 21\%$ at signal-to-noise ratios (SNR) of 20dB, respectively were developed [84]. In the context of AI-driven CVD care for Indigenous populations, Jeong et al. [85] pointed out that AI and ML-based predictive models could be robust solutions to CVD care rather than conventionally-used methods of care. In an experiment, Rajkumar et al. [86] echoed the superiority of IoT and DL-based methods in predicting CVD onset with an accuracy of 98.01%, boasting an error rate of 91.11% compared to other existing techniques.

5.2. Summary of Key Findings

The findings indicate that this unique model of care delivers substantial value, transforming the journey towards predictive and preventive CVD care and improving health outcomes for First Nations people. Further, adopting IoT/IoMT sensor-driven telehealth can reduce health inequalities and accelerate opportunities between urban and rural communities. The findings also emphasised the potential of wearable IoMT ECG and AI-driven telehealth systems to address chronic disease burdens in underserved communities and minimise health gaps between urban, rural, and remote regions.

The most consistent finding in this review highlighted the potential of integrating IoMT technology with telehealth ecosystems to provide evidencebased, safe, and reliable patient care for CVD. This indicates that wearable ECG and other biomedical flexible sensors offer significant advantages such as scalability, flexibility, low power consumption, and cost-effectiveness. This research suggests that wearables, IoT/IoMT electrocardiogram sensors, and AI-driven telehealth present potential opportunities for CVD diagnosis, prognosis, and management for rural and remote patients. Nevertheless, the review proposes that wearable IoMT ECG and AI-driven telehealth have the potential to transform CVD care for rural, regional, and remote aboriginal and non-aboriginal populations, providing accessible, cost-effective, and efficient smart healthcare solutions. In conclusion, this review points out that the long-term sustainability of the innovative IoT/IoMT sensor-driven telehealth care in various settings would be an enormous challenge for Australian rural healthcare infrastructure, combined with the complexity of the public healthcare administration.

In particular, the studies highlighted the disjointed nature of Australian healthcare data, the importance of accurate ECG signal classifications, and secured data privacy in facilitating the adoption of AI and big data in remote mHealth care settings. Further, AI-based predictive models emerged as robust solutions to address these issues, emphasising the superiority of wearable IoT-driven telehealth methods for CVD care dedicated towards remote Aboriginal peoples in Australia.

5.3. Barriers and Facilitators of Wearable IoT/IoMT sensors-driven Telehealth for CVD Care

Following the evidence from this review, for example, [19, 52, 12, 47, 21, 1, 75] we argue that the wearable IoT ECG sensors, AI-driven telehealth methods of care have had clear benefits for CVD care for rural, remote communities globally. A key question for future research to consider is: what are the barriers that prevent this life-saving technology from being implemented in rural Australia and similar settings?

This model of care has clinically been tested in various CVD conditions (adults, children, male, and female patients), and almost all types of cardiovascular disease have been screened and predicted remotely. This novel model of care provides patients and physicians with greater opportunities and flexibilities for CVD detection, early prediction, prevention, and management. This virtual model of care appears sound and could become a benchmark model to study AI-driven telehealth within the Aboriginal health domain. We believe the successful deployment of this innovative model of care will improve rural indigenous patients with CVDs and reduce health inequality among indigenous communities. Our claims are consistent with Calabria et al. [7]. CVDs are responsible for 21% of fatal diseases burden and the most prominent health disparity between aboriginal and non-aboriginal Australians [7]. Telehealth is suitable for serving broader rural and remote patients with CVD due to its technology-driven nature. Waller et al. [87] revealed that functional telehealth serving remote patients providing real-time consultation, diagnosis (e.g., echocardiogram), monitoring (e.g., EKG, glucose monitor and patients with congestive heart failure) and mentoring (e.g., another specialist observes and provide advice a remote real-time operation and virtual ICU). In a case study, Taylor et al. [88] echoed that A Children's Mercy Hospital (CHM) in Kansas City, MO, telemedicine dominantly support children in a variety of settings incorporating primary care, speciality care, pulmonary function tests, radiographs, and echocardiograms. This case study demonstrates positive public acceptance and demand for telemedicine in rural Missouri and Kansas and has led to a massive expansion, resulting in over 2000 outpatient encounters last year with a high growth rate exceeding over 40%. CHMs facilitated telemedicine now encompasses 27 paediatric specialities across four regional locations with additional expansions underway [88].

Similarly, Albahri et al. [19] commented that the wearable IoMT ECG Sensor, AI-driven telehealth, promises a vast improvement of services for remote care without incurring high medical costs. AI-based predictive aspects in the systems can assist in avoiding delays whilst timely medical treatment even before patients with CVD reach a severe condition [19]. Likewise, Zerna et al. [89] argue that multidisciplinary stroke expertise physicians are sufficiently unavailable in many rural areas, which makes delivering appropriate CVD (i.e., stroke) care in such areas a significant public health challenge. The heart of this novel method (i.e., wearable IoMT sensors, AI-driven telehealth) of care lies in its perspective on how it benefits both remote living care seekers (i.e., Aboriginal community) and urban-based care providers (i.e., physicians) and continues to serve them well.

To summarise, the contemporary literature discussed above on wearable IoMT sensors used for CVD care has had a narrow focus on clinical practices via AI-driven telehealth, particularly for rural patients. Insufficient attention has been directed towards conducting clinical trials using a large sample size. This inadequacy is reflected in the infrastructure facilities accessed and through evidence-based clinical practices. This review also exposes that the determinants of patients' acceptance, expectations, and satisfaction with this virtual care are still unidentified. What is lacking is an insight into how wearable IoMT ECG sensors collect bio-signal data and analyse them to arrive at a course of medical treatment via telehealth (medical decisions) for rural and remote communities. To point out this could drastically reduce fatalities, treatable illnesses, and related problems substantially.

From a compatibility standpoint, it is also unclear how these various MedTech technologies are compatible with traditional IoT/IoMT technologies and fit together, shaping a smart telehealth platform to create potential opportunities for rural and remote care. This indicates that research was undertaken haphazardly in recent decades. This drawback and limitations of current literature remind us that the AI-driven telehealth systems and underlying various services/facilities are not entirely recognised, even though the need is dire, and applications are readily available. Given these outcomes, we argue that health providers, policymakers, researchers, and stakeholders demand new knowledge about each aspect of this virtual care. The potential to bring health equality outcomes and reduce fatalities in treatable conditions in these communities is paramount. This new knowledge could also be shared between and across health industries strengthening service productivity, systems sustainability, and growth in Australia and globally.

6. Contribution and Managerial Implications

This study has a threefold contribution to health informatics literature. The first contribution is the investigation and identification of the adoption determinants of the wearable IoMT sensor-driven telehealth model for rural CVD care, which continues to be an under-researched area. To the authors' knowledge, this is a novel study that sheds light on the viability of adopting wearable IoMT ECG sensors within an AI-driven telehealth framework for Indigenous patients with CVD living in rural and remote Australia. This review demonstrates that there is a need for better care in rule and indigenous communities; that is, there is an inequality of access to good health-care because of the remoteness and inaccessibility of healthcare professionals. These inadequacies are evident in the existing research both in Australia and abroad.

Second, this review demonstrates the need for adopting a wearable IoMT sensor AI-driven telehealth model for real-time care appropriate for Aboriginal and non-Aboriginal patients with CVD living in rural and remote Australia and similar settings. The adoption of this technology widens telehealth scope in developed countries' rural contexts and could drastically improve health outcomes. Thus, we believe this research further broadens the

scope of AI-based health informatics research and provides helpful directions to health providers, policymakers, health authorities and stakeholders to integrate approaches to strengthen telehealth adaptability and sustainability for locational disadvantaged communities.

Third, this study makes a significant contribution to indigenous health, particularly by clarifying further how the various IoT/IoMT and sensor (MedTech) technologies could work together to build a novel telehealth model for remote CVD care in Australia and similar settings. Further, the long-term sustainability of the smart telehealth project is one of the most dominating challenges for rural healthcare infrastructure, alongside the complexity of the rural public healthcare administration. Failure to recognise these challenges could undermine their potential efficacy and years of hard work.

From a managerial perspective, the findings from the research have important implications for information systems, particularly in health informatics research. This review's findings illustrate that this care model has been clinically tested in various CVD conditions (adults, children, males, and female patients), and almost all types of CVD have been screened and predicted remotely. This novel model of care provides patients and physicians with greater opportunities and flexibilities for CVD detection, early prediction, prevention, and management. This virtual model of care appears sound and could become a benchmark model to study AI-driven telehealth within the Aboriginal health domain. This suggests that the successful deployment of this innovative model of care will improve rural indigenous patients with CVDs and reduce health inequality among indigenous communities. Given these discoveries, health providers and policymakers should design effective strategies, develop favourable policy guidelines, and implement Aboriginal CVD management plans for achieving goals.

7. Research Impact

The present study's underlying impacts are classified into four-dimensional categories: knowledge, novelty, Australian First Nations Peoples, and developed country. This study's ability contributed to defining, distinguishing, explaining, and interpreting wearable IoT/IoMT-ECG sensors and AI-driven telehealth adoption determinants in contexts of developed countries' rural, regional, and remote settings. It provides unique insights into rural Australian First Nations' CVD treatment regarding telehealth services. This study explored the adoption determinants of the AI-driven telehealth

model for CVD care. This study focused on adopting an AI-driven telehealth ecosystem for potentially applicable indigenous communities living in rural and remote Australia and similar settings. This research highlighted how this novel model of care provides patients and physicians with greater opportunities and flexibilities for CVD detection, early prediction, prevention, and disease management. This research identified wearable ECG sensors, such as a wrist-worn, dynamic watch that can detect cardiovascular signals for early diagnosis, predict patients' current CVD conditions, and manage cardiovascular diseases by early interventions.

This study introduced smart wearable IoT embedded MedTech, functional materials, network configurations, and bio-signal detection algorithms (i.e., AI and Machine Learning/Deep Learning) and their advantages. It emphasised how innovative telehealth synchronises digital and physical therapeutic modalities, reduces remote diagnostic hurdles, facilitates adaptable, comfortable, reliable, and economical healthcare interventions, and bridges inequality gaps between urban and rural health landscapes. Further, this research identified four areas of impact, namely (1) research-related (i.e., research problem, methods used, research management and communication), (2) policy (i.e., level of policy making, type, nature and policy networks), (3) service (i.e., health services, service management, quality of care and information systems) [90].

8. Limitations and Future Research

Nevertheless, several limitations should be considered when interpreting these research findings. This review used only four scientific databases for data extraction and synthesis. The future review should include additional databases to extract more data that may influence the broad view of the phenomenon. The future search should broaden by including other chronic disease risks and management using these technologies to observe the viability and effectiveness of this model of care on a large-scale adoption. Finally, our study did not undertake a formal quality assessment of the incorporated literature, thus constraining our ability to critically evaluate the sources utilised to substantiate our assertions.

9. Conclusion

One relatively unexplored research area involves the broad adoption of wearables IoMT sensors AI-driven telehealth focusing on rural and remote communities. Although wearables Internet of Medical Things electrocardiogram sensors, Artificial Intelligence-driven telehealth hold increased opportunities for cardiovascular disease diagnosis, prognosis, and management for rural and remote patients. Incorporating a large volume of research on the issues discussed in this review serves as a comprehensive guideline and list of the sources to lead the way in adopting this novel model of care for the Aboriginal communities living in rural and remote Australia.

The novelty of this study has advanced a pragmatic understanding of sustaining this cutting-edge model of care tackling high risks of CVD, causing deaths, challenges, and potential future directions of continuous monitoring and ubiquitous medical treatment via telehealth for underprivileged and vulnerable aboriginal communities. The study findings highlight the essential tools and resources that should be taken into consideration by the relevant authorities for extensive adoption. The government, health authorities, policymakers, health providers, and stakeholders are urged to work together. emphasising the implementation strategies for an initial pilot project as setting up a foundation before its widespread adoption in rural Australia and similar settings. The successful adoption of a wearable IoMT sensor-driven telehealth model of care could help reduce health inequalities affecting underprivileged rural/remote minorities. Rural regional and remote populations often have limited access to dedicated public medical facilities, and most lack private after-hours medical practitioners [34]. Hence, these locational disadvantages and sparsely populated regions should continue to be a priority [34].

Conflict of Interest: The authors declare no conflict of interest in this research.

Funding: The authors declare no funding for this research.

Data availability Statement: The authors confirm that no data were used in the research described in this article. All analyses, findings, and conclusions presented in this paper are based on the published literature and theoretical frameworks. No data sets, codes, or other materials are associated with this study.

References

[1] E. Haynes, R. Walker, A. G. Mitchell, J. Katzenellenbogen, H. D'Antoine, D. Bessarab, Decolonizing indigenous health: Generating a productive dialogue to eliminate rheumatic heart disease in australia, Social Science & Medicine 277 (2021) 113829.

- [2] J. Smylie, M. Firestone, The health of indigenous peoples, Social determinants of health: Canadian perspectives (2016) 434–469.
- [3] C. R. Valeggia, J. J. Snodgrass, Health of indigenous peoples, Annual Review of Anthropology 44 (2015) 117–135.
- [4] A. C. Carr, E. Spencer, H. Heenan, H. Lunt, M. Vollebregt, T. C. Prickett, Vitamin c status in people with types 1 and 2 diabetes mellitus and varying degrees of renal dysfunction: Relationship to body weight, Antioxidants 11 (2) (2022) 245.
- [5] C. W. Ong, E. B. Lee, M. P. Twohig, A meta-analysis of dropout rates in acceptance and commitment therapy, Behaviour research and therapy 104 (2018) 14–33.
- [6] B. Porykali, A. Davies, C. Brooks, H. Melville, M. Allman-Farinelli, J. Coombes, Effects of nutritional interventions on cardiovascular disease health outcomes in aboriginal and torres strait islander australians: A scoping review, Nutrients 13 (11) (2021) 4084.
- [7] B. Calabria, R. J. Korda, R. W. Lovett, P. Fernando, T. Martin, L. Malamoo, J. Welsh, E. Banks, Absolute cardiovascular disease risk and lipid-lowering therapy among aboriginal and torres strait islander australians, Medical Journal of Australia 209 (1) (2018) 35–41.
- [8] J. Banna, A. Bersamin, Community involvement in design, implementation and evaluation of nutrition interventions to reduce chronic diseases in indigenous populations in the us: a systematic review, International journal for equity in health 17 (1) (2018) 1–26.
- [9] N. Adelson, The embodiment of inequity: Health disparities in aboriginal canada, Canadian journal of public health 96 (2) (2005) S45–S61.
- [10] S. Crengle, J. N. Luke, M. Lambert, J. K. Smylie, S. Reid, J. Harré-Hindmarsh, M. Kelaher, Effect of a health literacy intervention trial on knowledge about cardiovascular disease medications among indigenous peoples in australia, canada and new zealand, BMJ open 8 (1) (2018) e018569.

Journal Pre-proof

- [11] M. L. Barnett, H. A. Huskamp, Telemedicine for mental health in the united states: Making progress, still a long way to go, Psychiatric Services 71 (2) (2020) 197–198.
- [12] C. Beach, S. Krachunov, J. Pope, X. Fafoutis, R. J. Piechocki, I. Craddock, A. J. Casson, An ultra low power personalizable wrist worn ecg monitor integrated with iot infrastructure, IEEE Access 6 (2018) 44010– 44021.
- [13] E. Piotrowicz, M. F. Piepoli, T. Jaarsma, E. Lambrinou, A. J. Coats, J.-P. Schmid, U. Corrà, P. Agostoni, K. Dickstein, P. M. Seferović, S. Adamopoulos, P. P. Ponikowski, Telerehabilitation in heart failure patients: The evidence and the pitfalls, International Journal of Cardiology 220 (2016) 408–413.
- [14] S. Raj, S. Luthra, K. C. Ray, Development of handheld cardiac event monitoring system, IFAC-PapersOnLine 48 (4) (2015) 71–76.
- [15] S. Sanamdikar, S. Hamde, V. Asutkar, Classification and analysis of cardiac arrhythmia based on incremental support vector regression on iot platform, Biomedical Signal Processing and Control 64 (2021) 102324.
- [16] L. Brazionis, A. Jenkins, A. Keech, C. Ryan, S.-E. Bursell, T. S. Group, An evaluation of the telehealth facilitation of diabetes and cardiovascular care in remote australian indigenous communities:-protocol for the telehealth eye and associated medical services network [teamsnet] project, a pre-post study design, BMC health services research 17 (2017) 1–12.
- [17] O. Gibson, K. Lisy, C. Davy, E. Aromataris, E. Kite, C. Lockwood, D. Riitano, K. McBride, A. Brown, Enablers and barriers to the implementation of primary health care interventions for indigenous people with chronic diseases: a systematic review, Implementation Science 10 (1) (2015) 71.
- [18] O. Adegbija, W. Hoy, Z. Wang, Prediction of cardiovascular disease risk using waist circumference among aboriginals in a remote australian community, BMC Public Health 15 (1) (2015) 1–9.
- [19] A. S. Albahri, J. K. Alwan, Z. K. Taha, S. F. Ismail, R. A. Hamid, A. Zaidan, O. S. Albahri, B. Zaidan, A. Alamoodi, M. Alsalem, Iot-
based telemedicine for disease prevention and health promotion: Stateof-the-art, Journal of Network and Computer Applications 173 (2021) 102873.

- [20] U. Albalawi, S. Joshi, Secure and trusted telemedicine in internet of things iot, in: 2018 IEEE 4th World Forum on Internet of Things (WF-IoT), IEEE, 2018, pp. 30–34.
- [21] H. Hamil, Z. Zidelmal, M. S. Azzaz, S. Sakhi, R. Kaibou, S. Djilali, D. Ould Abdeslam, Design of a secured telehealth system based on multiple biosignals diagnosis and classification for iot application, Expert Systems (2021) e12765.
- [22] A. M. Koya, P. Deepthi, Plug and play self-configurable iot gateway node for telemonitoring of ecg, Computers in Biology and Medicine 112 (2019) 103359.
- [23] V. Morgan, M. Birtus, A. Zauskova, Medical internet of things-based healthcare systems, wearable biometric sensors, and personalized clinical care in remotely monitoring and caring for confirmed or suspected covid-19 patients, American Journal of Medical Research 8 (1) (2021) 81–90.
- [24] A. Riley, E. Nica, Internet of things-based smart healthcare systems and wireless biomedical sensing devices in monitoring, detection, and prevention of covid-19, American Journal of Medical Research 8 (2) (2021) 51–64.
- [25] J. Sawyer, Wearable internet of medical things sensor devices, artificial intelligence-driven smart healthcare services, and personalized clinical care in covid-19 telemedicine, American Journal of Medical Research 7 (2) (2020) 71–77.
- [26] M. C. Lu, The future of maternal and child health, Maternal and child health journal 23 (2019) 1–7.
- [27] D. Medin-Eastwood, I. Podhorska, P. Kubala, Healthcare big data systems, wearable medical devices, and remote patient care, American Journal of Medical Research 6 (1) (2019) 49–54.
- [28] R. Dwivedi, D. Mehrotra, S. Chandra, Potential of internet of medical things (iomt) applications in building a smart healthcare system: A

systematic review, Journal of Oral Biology and Craniofacial Research (2021).

- [29] N. A. Bajao, J.-a. Sarucam, Threats detection in the internet of things using convolutional neural networks, long short-term memory, and gated recurrent units, Mesopotamian journal of cybersecurity 2023 (2023) 22– 29.
- [30] F. Dahlqvist, M. Patel, A. Rajko, J. Shulman, Growing opportunities in the internet of things, McKinsey, July (2019).
- [31] P.-T. Chen, C.-L. Lin, W.-N. Wu, Big data management in healthcare: adoption challenges and implications, International Journal of Information Management (2020) 102078.
- [32] W. H. Organization, et al., Global strategy on digital health 2020-2025., World Health Organization, 2021.
- [33] K. Byerly, L. Vagner, I. Grecu, G. Grecu, G. Lăzăroiu, Real-time big data processing and wearable internet of medical things sensor devices for health monitoring, American Journal of Medical Research 6 (2) (2019) 67–72.
- [34] K. M. Zobair, L. Sanzogni, L. Houghton, M. Z. Islam, Forecasting care seekers satisfaction with telemedicine using machine learning and structural equation modeling, Plos one 16 (9) (2021) e0257300.
- [35] J. Lin, R. Fu, X. Zhong, P. Yu, G. Tan, W. Li, H. Zhang, Y. Li, L. Zhou, C. Ning, Wearable sensors and devices for real-time cardiovascular disease monitoring, Cell Reports Physical Science 2 (8) (2021) 100541.
- [36] R. Disler, K. Glenister, J. Wright, Rural chronic disease research patterns in the united kingdom, united states, canada, australia and new zealand: a systematic integrative review, BMC Public Health 20 (1) (2020) 1–7.
- [37] X. Hua, R. McDermott, T. Lung, M. Wenitong, A. Tran-Duy, M. Li, P. Clarke, Validation and recalibration of the framingham cardiovascular disease risk models in an australian indigenous cohort, European journal of preventive cardiology 24 (15) (2017) 1660–1669.

- [38] A. Liberati, D. G. Altman, J. Tetzlaff, C. Mulrow, P. C. Gøtzsche, J. P. Ioannidis, M. Clarke, P. J. Devereaux, J. Kleijnen, D. Moher, The prisma statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration, Journal of clinical epidemiology 62 (10) (2009) e1–e34.
- [39] C. Kraef, M. van der Meirschen, C. Free, Digital telemedicine interventions for patients with multimorbidity: a systematic review and metaanalysis, BMJ open 10 (10) (2020) e036904.
- [40] Y. Jin, H. Yu, Y. Zhang, N. Pan, M. Guizani, Predictive analysis in outpatients assisted by the internet of medical things, Future Generation Computer Systems 98 (2019) 219–226.
- [41] E. Grooby, J. He, J. Kiewsky, D. Fattahi, L. Zhou, A. King, A. Ramanathan, A. Malhotra, G. A. Dumont, F. Marzbanrad, Neonatal heart and lung sound quality assessment for robust heart and breathing rate estimation for telehealth applications, IEEE Journal of Biomedical and Health Informatics (2020).
- [42] F. Rashid, A. Khalid, Design and implementation of real-time electrocardiogram monitoring system for telemedicine services, Indian Journal of Science and Technology 12 (2019) 16.
- [43] V. Randazzo, J. Ferretti, E. Pasero, Ecg watch: a real time wireless wearable ecg, in: 2019 IEEE International Symposium on Medical Measurements and Applications (MeMeA), IEEE, 2019, pp. 1–6.
- [44] F. Wang, Z. Li, X. Guo, W. Liao, Quality prediction and control of cable harness wiring using extension theory and a backpropagation neural network, Advances in Mechanical Engineering 12 (5) (2020) 1687814020923103.
- [45] A. Vallée, A. Cinaud, V. Blachier, H. Lelong, M. E. Safar, J. Blacher, Coronary heart disease diagnosis by artificial neural networks including aortic pulse wave velocity index and clinical parameters, Journal of Hypertension 37 (8) (2019) 1682–1688.
- [46] S. Shomaji, P. Dehghanzadeh, A. Roman, D. Forte, S. Bhunia, S. Mandal, Early detection of cardiovascular diseases using wearable ultrasound device, IEEE Consumer Electronics Magazine 8 (6) (2019) 12–21.

- [47] H. De Canniére, F. Corradi, C. J. Smeets, M. Schoutteten, C. Varon, C. Van Hoof, S. Van Huffel, W. Groenendaal, P. Vandervoort, Wearable monitoring and interpretable machine learning can objectively track progression in patients during cardiac rehabilitation, Sensors 20 (12) (2020) 3601.
- [48] Z. Yang, Q. Zhou, L. Lei, K. Zheng, W. Xiang, An iot-cloud based wearable ecg monitoring system for smart healthcare, Journal of medical systems 40 (12) (2016) 286.
- [49] P. Balsam, P. Lodziński, A. Tymińska, K. Ozierański, Januszkiewicz, R. Główczyńska, K. Wesołowska, M. Peller, R. Pietrzak, T. Ksiżczyk, Study design and rationale for biomedical shirt-based electrocardiography monitoring in relevant clinical situations: Ecg-shirt study, Cardiology journal 25 (1) (2018) 52–59.
- [50] A. Malekian Borujeni, M. Fathy, N. Mozayani, A hierarchical, scalable architecture for a real-time monitoring system for an electrocardiography, using context-aware computing, Journal of Biomedical Informatics 96 (2019) 103251.
- [51] M. A. Al-Alusi, E. Ding, D. D. McManus, S. A. Lubitz, Wearing your heart on your sleeve: the future of cardiac rhythm monitoring, Current cardiology reports 21 (2019) 1–11.
- [52] N. Baghel, M. K. Dutta, R. Burget, Automatic diagnosis of multiple cardiac diseases from pcg signals using convolutional neural network, Computer Methods and Programs in Biomedicine 197 (2020) 105750.
- [53] M. P. McRae, B. Bozkurt, C. M. Ballantyne, X. Sanchez, N. Christodoulides, G. Simmons, V. Nambi, A. Misra, C. S. Miller, J. L. Ebersole, Cardiac scorecard: A diagnostic multivariate index assay system for predicting a spectrum of cardiovascular disease, Expert systems with applications 54 (2016) 136–147.
- [54] J. M. Pevnick, K. Birkeland, R. Zimmer, Y. Elad, I. Kedan, Wearable technology for cardiology: an update and framework for the future, Trends in cardiovascular medicine 28 (2) (2018) 144–150.

- [55] D. Florez, J. Sepulveda, Blooxy: On a non-invasive blood monitor for the iot context, in: 2017 30th IEEE International System-on-Chip Conference (SOCC), IEEE, 2017, pp. 29–34.
- [56] D. Tsay, C. Patterson, From machine learning to artificial intelligence applications in cardiac care: Real-world examples in improving imaging and patient access, Circulation 138 (22) (2018) 2569–2575.
- [57] S. A. Bini, Artificial intelligence, machine learning, deep learning, and cognitive computing: What do these terms mean and how will they impact health care?, The Journal of Arthroplasty 33 (8) (2018) 2358– 2361.
- [58] C. Krittanawong, K. W. Johnson, R. S. Rosenson, Z. Wang, M. Aydar, U. Baber, J. K. Min, W. W. Tang, J. L. Halperin, S. M. Narayan, Deep learning for cardiovascular medicine: a practical primer, European heart journal 40 (25) (2019) 2058–2073.
- [59] O. Faust, N. Lei, E. Chew, E. J. Ciaccio, U. R. Acharya, A smart service platform for cost efficient cardiac health monitoring, International journal of environmental research and public health 17 (17) (2020) 6313.
- [60] F. Ma, J. Zhang, W. Liang, J. Xue, Automated classification of atrial fibrillation using artificial neural network for wearable devices, Mathematical Problems in Engineering 2020 (2020).
- [61] C. Power, Alex brown: fighting for the health of indigenous australians, The Lancet Diabetes & Endocrinology 9 (3) (2021) 142.
- [62] S. S. Heraganahally, T. Howarth, L. Sorger, H. Ben Saad, Sex differences in pulmonary function parameters among indigenous australians with and without chronic airway disease, PLOS ONE 17 (2) (2022) e0263744.
- [63] A. Brown, W. Walsh, T. Lea, A. Tonkin, What becomes of the broken hearted? coronary heart disease as a paradigm of cardiovascular disease and poor health among indigenous australians, Heart, Lung and Circulation 14 (3) (2005) 158–162.
- [64] A. W. Gaffney, L. Hawks, A. C. White, S. Woolhandler, D. Himmelstein, D. C. Christiani, D. McCormick, Health care disparities across

the urban-rural divide: a national study of individuals with copd, The Journal of Rural Health 38~(1)~(2022)~207-216.

- [65] M. Donati, A. Celli, A. Ruiu, S. Saponara, L. Fanucci, A telemedicine service system exploiting bt/ble wireless sensors for remote management of chronic patients, Technologies 7 (1) (2019) 13.
- [66] L. H. Schwamm, N. Chumbler, E. Brown, G. C. Fonarow, D. Berube, K. Nystrom, R. Suter, M. Zavala, D. Polsky, K. Radhakrishnan, Recommendations for the implementation of telehealth in cardiovascular and stroke care: a policy statement from the american heart association, Circulation 135 (7) (2017) e24–e44.
- [67] K. Butten, N. W. Johnson, K. K. Hall, M. Toombs, N. King, K.-A. F. O'Grady, Yarning about oral health: perceptions of urban australian aboriginal and torres strait islander women, BMC Oral Health 20 (2020) 1–12.
- [68] Y. Zhu, X. Gu, C. Xu, Effectiveness of telemedicine systems for adults with heart failure: a meta-analysis of randomized controlled trials, Heart failure reviews 25 (2) (2020) 231–243.
- [69] E. M. Cronin, E. A. Ching, N. Varma, D. O. Martin, B. L. Wilkoff, B. D. Lindsay, Remote monitoring of cardiovascular devices: a time and activity analysis, Heart Rhythm 9 (12) (2012) 1947–1951.
- [70] A. Martinez-Millana, C. Palao, C. Fernandez-Llatas, P. de Carvalho, A. M. Bianchi, V. Traver, Integrated iot intelligent system for the automatic detection of cardiac variability, in: 2018 40th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), IEEE, pp. 5798–5801.
- [71] O. S. Albahri, A. Zaidan, B. Zaidan, M. Hashim, A. S. Albahri, M. Alsalem, Real-time remote health-monitoring systems in a medical centre: A review of the provision of healthcare services-based body sensor information, open challenges and methodological aspects, Journal of medical systems 42 (2018) 1–47.
- [72] M. Scheffler, E. Hirt, Wearable devices for telemedicine applications, Journal of telemedicine and telecare 11 $(1_s uppl)(2005)11 -14$.

- [73] G. Cugliari, S. Benevenuta, S. Guarrera, C. Sacerdote, S. Panico, V. Krogh, R. Tumino, P. Vineis, P. Fariselli, G. Matullo, Improving the prediction of cardiovascular risk with machine-learning and dna methylation data, in: 2019 IEEE Conference on Computational Intelligence in Bioinformatics and Computational Biology (CIBCB), IEEE, pp. 1–4.
- [74] M. Cousin, T. Castillo-Hi, G. H. Snyder, Devices and diseases: How the iot is transforming medtech, Retrieved September, 2017 (2015).
- [75] R. D. Kindle, O. Badawi, L. A. Celi, S. J. C. c. c. Sturland, Intensive care unit telemedicine in the era of big data, artificial intelligence, and computer clinical decision support systems 35 (3) (2019) 483–495.
- [76] Y. Liu, H. Huang, F. Xiao, R. Malekian, W. Wang, Classification and recognition of encrypted eeg data based on neural network, Journal of Information Security and Applications 54 (2020) 102567.
- [77] S. Majumder, T. Mondal, M. J. Deen, Wearable sensors for remote health monitoring, Sensors 17 (1) (2017) 130.
- [78] X.-L. Shen, Y.-J. Li, Y. Sun, Wearable health information systems intermittent discontinuance: A revised expectation-disconfirmation model, Industrial Management & Data Systems (2018).
- [79] S. Chen, J. Qi, S. Fan, Z. Qiao, J. C. Yeo, C. T. Lim, Flexible wearable sensors for cardiovascular health monitoring, Advanced Healthcare Materials 10 (17) (2021) 2100116.
- [80] V. Blake, L. Jorm, J. Yu, A. Lee, B. Gallego, S.-Y. Ooi, The cardiac analytics and innovation (cardiacai) data repository: An australian data resource for translational cardiovascular research, arXiv preprint arXiv:2304.09341 (2023).
- [81] A. Deniz-Garcia, H. Fabelo, A. J. Rodriguez-Almeida, G. Zamora-Zamorano, M. Castro-Fernandez, M. d. P. Alberiche Ruano, T. Solvoll, C. Granja, T. R. Schopf, G. M. Callico, C. Soguero-Ruiz, A. M. Wägner, Quality, usability, and effectiveness of mhealth apps and the role of artificial intelligence: Current scenario and challenges, J Med Internet Res 25 (2023) e44030.

- [82] L. Sun, Y. Wang, Z. Qu, N. N. Xiong, Beatclass: A sustainable ecg classification system in iot-based ehealth, IEEE Internet of Things Journal 9 (10) (2022) 7178–7195. doi:10.1109/JIOT.2021.3108792.
- [83] W. Hu, F. S. Yii, R. Chen, X. Zhang, X. Shang, K. Kiburg, E. Woods, A. Vingrys, L. Zhang, Z. Zhu, et al., A systematic review and meta-analysis of applying deep learning in the prediction of the risk of cardiovascular diseases from retinal images, Translational Vision Science & Technology 12 (7) (2023) 14–14.
- [84] A. John, A. Padinjarathala, E. Doheny, B. Cardiff, D. John, An evaluation of ecg data fusion algorithms for wearable iot sensors, Information Fusion 96 (2023) 237–251.
- [85] K. Jeong, A. R. Mallard, L. Coombe, J. Ward, Artificial intelligence and prediction of cardiometabolic disease: Systematic review of model performance and potential benefits in indigenous populations, Artificial Intelligence in Medicine 139 (2023) 102534.
- [86] G. Rajkumar, T. Gayathri Devi, A. Srinivasan, Heart disease prediction using iot based framework and improved deep learning approach: Medical application, Medical Engineering & Physics 111 (2023) 103937.
- [87] M. Waller, C. Stotler, Telemedicine: a primer, Current allergy and asthma reports 18 (2018) 1–9.
- [88] L. Taylor, H. Capling, J. M. Portnoy, Administering a telemedicine program, Current allergy and asthma reports 18 (2018) 1–7.
- [89] C. Zerna, T. Jeerakathil, M. D. Hill, Telehealth for remote stroke management, Canadian Journal of Cardiology (2017).
- [90] S. Kuruvilla, N. Mays, A. Pleasant, G. Walt, Describing the impact of health research: a research impact framework, BMC health services research 6 (1) (2006) 1–18.
- [91] C. Liu, X. Zhang, L. Zhao, F. Liu, X. Chen, Y. Yao, J. Li, Signal quality assessment and lightweight qrs detection for wearable ecg smartvest system, IEEE Internet of Things Journal 6 (2) (2019) 1363–1374.

Appendix A. Literature Search results on Wearable IoT/IoMT, ECG sensors, AI-driven Telehealth for CVD care.

44

Table A.1: Literature Search results on Wearable IoT/IoMT, ECG sensors, AI-driven Telehealth for CVD care.

for biomedical algorithm: the shirt-based shapiro-Wilk electrocardiography electrocardiography electrocardiography (ECG) monitoring of patients with CVD in monitoring in software study. SAS(R) (ECG) monitoring of patients with CVD in different clinical situations: software version 9.4. Nuubo ECG (nECG) situations: ECG-shirt study. ECG-shirt captures the electrocardiographic signal via textile electrocardiographic sidenteelectrocardiographic s	ID Year Author Country Study Methods Accuracy (%) f Proposal Outcome Measures Stud- Extud- Outcome Measures Stud- Stud- 0 ies Accuracy (%) Frankythmia 1 2018 [49] Poland Study design and rationale detecting utility of biomedical and rationale detecting utility of biomedical shirt-based Shapiro-Wilk electrocardiography electrocardiog test with (ECG) monitoring of raphy software situations employing the situations: software situations employing the study: study: systems. The proposed
--	--

ge	curacy@%jectives and	Outcome Measures		. In the proposed model,	a patient's vital signs	are measured using a	body sensor network	and sent to an	intelligent health care	domain. Empirical	findings confirm a	significant improvement	of 70% in response time	and scalability	compared to the	state-of-the-art	techniques. This	improvement is about	30% compared to Fog	computing approaches.	Continued on next page	
ious pa	\mathbf{Acc}			202									V									
inued from previ	Methods			Wireless body	area networks,	Global	Positioning	System, Blood	$\mathbf{Pressure},$	survey	systems,	activity	monitors, bod	temperature,	and ECG	signal detector						
Table A.1 $-$ cont	ry Study	$\operatorname{Proposal}$		A hierarchical,	scalable and	real-time	monitoring	system for	electrocardiog-	raphy using	context-aware	computing.										
	Count	of	Stud- ies	Iran																		
	Author			50]																		
	D Year 7			2019 [
				2																		

	acy016jectives and	Outcome Measures			This study configured	several groups of meanable FCC concore	comprising of AlivCor	Kardia devices, Apple	Watch Series 4, and	several others.	Continued on next page	
us page	Accura				92%	speci- ficity	90%	sensi-	tivity			
inued from previo	Methods				Arterial	Fibrillation	algorithms.)				
Table A.1 $-$ cont	ry Study	$\operatorname{Proposal}$			Wearing Your	Heart on Your Sleave: the	Future of	Cardiac	Rhythm	Monitoring.		
	Count	of	Stud-	ies	USA							
	Author				[51]							
	Year				2019 [
	H				ŝ							

bage	ccuracy@kjjectives and Outcome Measures	3.60% This study designed a phonocardiogram for automated and real-time cardiac disease. The authors used CNN for biosignal data classifications captured by PCG sensors. The authors also highlighted that PCG is compatible with any computing device, single board computing processors, android handheld devices.	Continued on next page	
Table A.1 – continued from previous pa	ry Study Methods Ac Proposal	Automatic CNN 98.0 diagnosis of multiple cardiac diseases from PCG signals using Convolutional Neural Neural Network (CNN).		
	ID Year Author Counti of Stud- ies	4 2020 [52] India		

ous page	Accuracy@bjbctives and	Outcome Measures			AUC=0.840àrdiac ScoreCard is a	for dis- multivariate index assay	ease vs. system for early	non- detection and frequent	case monitoring of	and traditional risk factors	0.9412 along with novel	for biomarkers for patients	cardiac with CVDs.	well-	ness vs.	HF.	Continued on next page
inued from previo	Methods				Lasso Logistic	regression.											
Table A.1 – cont	Country Study	of Proposal	Stud-	ies	USA Cardiac	ScoreCard: A	diagnostic	multivariate	index assay	system for	predicting a	spectrum of	cardiovascular	disease.			
	ID Year Author	5			5 2016 [53] 1												

49

page	ccuracy whetives and	Outcome Measures		5% Creating value in	chronic care for	Aboriginal populations	via telehealth, the	authors surveyed	proposing a telehealth	model for remote and	very remote indigenous	patients with	cardiovascular disease	and Diabetes.	Continued on next page
inued from previous	Methods P			Analysis of 9	covariance	(ANCOVA).									2
Table A.1 – conti	Country Study	of Proposal	Stud- ies	Australia Evaluation of	telehealth	facilitation of	Diabetes and	cardiovascular	care in	Indigenous	communities.				
	ID Year Author			6 2017 [16]											

age	curacy@bijectives and	Outcome Measures			% This study proposed	IoT integrated	electrocardiogram	sensor to monitor	different forms of beat	arrhythmias for	identifying heart	problems. The proposed	algorithm successfully	predicted cardiac	arrhythmias with 98%	accuracy.	Continued on next page
revious p ^a	$\mathbf{A}_{\mathbf{C}}$				1 989												
inued from p	Methods				Incrementa	$\operatorname{Support}$	Vector	Regression.									
Table A.1 $-$ cont	ountry Study	$\operatorname{Proposal}$	-pn		dia Classification f	cardiac	arrhythmia	based on	incremental	support vector	regression on	IoT platform.					
	uthor Cc	of	Sti	ies	5] Inc												
) Year A				2021 [1!												
	H				4												

Pronosed a wearable	IoT Wrist-Worn ECG	sensor to monitor home-based patients	with CVDs in the UK. This wearable offers	low-power consumptions	for real-time heart rate	variability monitoring.	Proposed an IoT-based	wearable 12-lead ECG	SmartVest system for	the early CVD	detection. The proposed	method can efficiently	deal with the trade-off	between accepting good	and rejecting poor	quality ECGs.	Continued on next page	
10%	vari-	ance of the	mean.				F1	score >	99.5%									
Flavihla tavtila	electrodes into	smart clothing.					Signal Quality	Assessment	and QRS	detection	algorithms							
An Illtra-Low	Power	Personalised Wrist Worn	ECG Monitor Integrated	With IoT	Infrastructure.		a Signal Quality	Assessment	and	Lightweight	QRS Detection	for Wearable	ECG	SmartVest.				
ies 11K	40						China											
2018 [19]	[7T] 0107						2019 [91]											
	ies ies 1018 [19] IIIZ An IIItra I arr Elavibla tartila 1002 Duanaad a uraarabla	8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Power electrodes into vari- IoT Wrist-Worn ECG	 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Power electrodes into vari- IoT Wrist-Worn ECG Personalised smart clothing. ance sensor to monitor Wrist Worn of the home-based patients 	and Power Flexible textile 10% Proposed a wearable 8<2018	Number Source ies An Ultra-Low Flexible textile 10% Proposed a wearable 8<2018	Note Flexible textile 10% Proposed a wearable 8<2018	Number Number ies 8<2018	ies 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 8<2018 [12]	Source Source Sector 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 9 2019 [91] China Signal Quality F1 Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality Signal Quality Signal Quality Signal Quality F1 Proposed a wearable	succession Signal Quality An Ultra-Low Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality Signal Quality Fit Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality Signal Quality Fit Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality Signal Quality Signal Quality Fit Proposed an IoT-based 9 2019 [91] China Signal Quality Signal Quality Fit Proposed an IoT-based 9 2019 [91] China Signal Quality Signal Quality Fit Proposed an IoT-based 9 2019 [91] Signal Quality Signal Quality Fit Proposed an IoT-based 9 2019 [91] Signal Quality Signal Quality Fit Proposed an IoT-based	source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality F1 Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality F1 Proposed an IoT-based and IoT-	source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 9 2019 [91] China Signal Quality F1 Proposed a wearable 9 2019 [91] China Signal Quality F1 Proposed an IoT-based 9 2019 [91] China Signal Quality F1 Proposed an IoT-based 9 2019 [91] China Signal Quality Signal Quality F1 Proposed an IoT-based 1 Assessment Assessment Socret wearable 12-lead ECG 1 Assessment Assessment score wearable 12-lead ECG 1 Assessment Assessment score wearable 12-lead ECG 1 Assessment and QRS 99.5% SmartVest system for 1 Assessment and QRS 99.5% SmartVest system for 1 Assessment and QRS 99.5% SmartVest system for	source Second	Source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality Fl Proposed an understom 9 2019 [91] China Signal Quality Signal Quality Fl Proposed an IoT-based and the and trafte 9 2019 [91] China Signal Quality Signal Quality Fl Proposed an IoT-based and the and trafte 9 2019 [91] China Signal Quality Signal Quality Fl Proposed an IoT-based and the early CVD 9 2019 [91] China Signal Quality Flor Proposed an IoT-based and the early CVD 9 2019 [91] China Signal Quality Flor Proposed an IoT-based and the early CVD 10 China Signal Quality Signal Quality Flor Proposed an IoT-based and the early CVD 10 China Signal Quality Signal Quality Flor Proposed and IoT-based and the early CVD <td>source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Personalised smart clothing. ance sensor to monitor sensor to monitor Wrist Worn Wrist Worn of the home-based patients ECG Monitor Integrated with CVDs in the UK. Integrated Mrist Worn of the home-based patients Writh IoT Integrated with CVDs in the UK. Integrated Mrith IoT nean. with CVDs in the UK. Integrated Mrith IoT Assessment score > 9 2019 [91] China Signal Quality 10 Signal Quality F1 Proposed an IoT-based Assessment and QRS 99.5% SmartVest system for Lightweight detection and QRS 99.5% ORS Detection algorithms detection. The proposed FCG Monitor and QRS 99.5% SmartVest. Assessment score > SmartVest. Assessment score ></td> <td>source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed 9 2019 [91] China Signal Quality F1 Proposed 9 2019 [91] Oh</td> <td>source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Personalised smart clothing. ance sensor to monitor wrist-Worn ECG Personalised smart clothing. ance sensor to monitor Wrist Worn Wrist Worn of the home-based patients Writh IoT mean. This wearable offers low-power consumptions Writh IoT Integrated writh CVDs in the UK. Writh IoT mean. This wearable offers Vith IoT mean. for real-time heart rate Vith IoT Assessment score > smartVest system for Journal Joulity F1 Proposed an IoT-based Assessment Assessment score > smartVest system for Assessment Assessment score > smartVest system for Lightweight detection and QRS 99.5% SmartVest system for CG Assessment algorithms detection. The proposed Rection Assessment score > score off More Assessment and QRS 99.5% between accepting god</td> <td>Source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Power electrodes into vari- IoT Wrist-Worn ECG Personalised smart clothing: ance sensor to monitor Wrist Worn Wrist Worn of the home-based patients Wrist IoT mean. with CVDs in the UK. Integrated with IoT mean. with CVDs in the UK. With IoT mean. with CVDs in the UK. Infrastructure. Mith IoT mean. with CVDs in the UK. 9 2019 [91] China Signal Quality F1 Proposed an IoT-based Assessment and QRS 99.5% SmartVest system for variability monitoring. 9 2019 [91] China Signal Quality F1 Proposed an IoT-based Assessment assessment score warable I2-lead ECG and QRS SinstVest system for 1 Assessment and QRS 99.5% SmartVest system for beto</td>	source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Personalised smart clothing. ance sensor to monitor sensor to monitor Wrist Worn Wrist Worn of the home-based patients ECG Monitor Integrated with CVDs in the UK. Integrated Mrist Worn of the home-based patients Writh IoT Integrated with CVDs in the UK. Integrated Mrith IoT nean. with CVDs in the UK. Integrated Mrith IoT Assessment score > 9 2019 [91] China Signal Quality 10 Signal Quality F1 Proposed an IoT-based Assessment and QRS 99.5% SmartVest system for Lightweight detection and QRS 99.5% ORS Detection algorithms detection. The proposed FCG Monitor and QRS 99.5% SmartVest. Assessment score > SmartVest. Assessment score >	source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable 9 2019 [91] China Signal Quality Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed an Ior-based 9 2019 [91] China Signal Quality F1 Proposed 9 2019 [91] China Signal Quality F1 Proposed 9 2019 [91] Oh	source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Personalised smart clothing. ance sensor to monitor wrist-Worn ECG Personalised smart clothing. ance sensor to monitor Wrist Worn Wrist Worn of the home-based patients Writh IoT mean. This wearable offers low-power consumptions Writh IoT Integrated writh CVDs in the UK. Writh IoT mean. This wearable offers Vith IoT mean. for real-time heart rate Vith IoT Assessment score > smartVest system for Journal Joulity F1 Proposed an IoT-based Assessment Assessment score > smartVest system for Assessment Assessment score > smartVest system for Lightweight detection and QRS 99.5% SmartVest system for CG Assessment algorithms detection. The proposed Rection Assessment score > score off More Assessment and QRS 99.5% between accepting god	Source Flexible textile 10% Proposed a wearable 8 2018 [12] UK An Ultra-Low Flexible textile 10% Proposed a wearable Power electrodes into vari- IoT Wrist-Worn ECG Personalised smart clothing: ance sensor to monitor Wrist Worn Wrist Worn of the home-based patients Wrist IoT mean. with CVDs in the UK. Integrated with IoT mean. with CVDs in the UK. With IoT mean. with CVDs in the UK. Infrastructure. Mith IoT mean. with CVDs in the UK. 9 2019 [91] China Signal Quality F1 Proposed an IoT-based Assessment and QRS 99.5% SmartVest system for variability monitoring. 9 2019 [91] China Signal Quality F1 Proposed an IoT-based Assessment assessment score warable I2-lead ECG and QRS SinstVest system for 1 Assessment and QRS 99.5% SmartVest system for beto

age	ccuracy@bjectives and	Outcome Measures			.7% Discusses the	characteristics,	applicability, and	telemonitoring tools to	provide optimal	long-term management	for heart failure patients	unable to attend	traditional	rehabilitation programs.	Continued on next page	
previous p	Ac				d. 94											,Q`
ontinued from I	Methods				ationUnspecifie	lre										
Table A.1 – c	ry Study	$\operatorname{Proposal}$			Telerehabilit	in heart failt	patients.									
	Countr	of	Stud-	ies	Poland											
	Author				[13]											
	ID Year				$10 \ 2016$											

ous page	Accuracy (Ki)ectives and	Outcome Measures			92.81%, handheld arrhythmias speci- detection ECG systems ficity for CVD diagnosis in 92.68% India. and pre- cision 92.42%. Continued on next page	
inued from previo	Methods				Component Analysis and Feed-forward Neural Networks.	
Table A.1 $-$ conti	Country Study	f Proposal	tud-	Se	of Handheld Cardiac Event Monitoring System.	
	ID Year Author C	0	S	ie		

us page	Accuracy (bi) betives and	Outcome Measures			R wave Proposed an	0.68 in- IoT-cloud-based ECG	dicates sensor for real-time	health patients with CVDs	heart. monitoring. The	proposed system is	reliable in collecting and	displaying real-time	ECG signal data, which	may contribute to the	diagnosis of certain	patients with CVDs.	Continued on next page
nued from previc	Methods				Unspecified.												
Table A.1 $-$ conti	Country Study	of Proposal	Stud-	ies	China An IoT-cloud	Based	Wearable ECG	Monitoring	System for	Smart	Healthcare.						
	ID Year Author				12 2016 [48]												

e	uracy@%jectives and	Outcome Measures			Proposed a	multi-parameter	wearable sensor for	follow-up cardiac	rehabilitation patients	in Belgium. Applied a	interpretable ML	approach to visualise	the relationships	between sensor-derived	biomarkers and their	ability to monitor	remote patients.	Continued on next page	
ious pag	Acc				66%														
inued from previ	Methods				Support	Vector	Machine	(SVM).											
Table A.1 $-$ cont	ry Study	$\operatorname{Proposal}$			1 Wearable	monitoring	and	Interpretable	ML to track	progression in	patients during	cardiac	rehabilitation.						
	Countr	of	Stud-	ies	Belgium														
	Author				47]														
	ID Year				13 2020														

56

			TADIE A.1 – COIII	inuea trom previ	ious page	
ID Year A	uthor	Country	y Study	Methods	Accuracy@%jectives and	
		of	$\operatorname{Proposal}$		Outcome Measures	
		Stud- ies				
$14 \ 2019 \ [46]$]]	USA	Early	Wearable	UnspecifieRroposed a wearable	
	,		detection of	imaging	ultrasound imaging	
			CVD using	system	device for the early	
			wearable	algorithm.	detection of CVDs	
			ultrasound		through a quantitative	
			device.		carotid artery	
					monitoring system in	
					the USA. The	
					distinctive features of	
					the device include	
					portability, accuracy,	
					ease of use, and	
					cost-effectiveness. The	
					authors outlined a set of	
					diagnostic tools	
					including CT heart	
					scans, chest X-rays,	
					blood tests, cardiac	
					catheterisation, heart	
					MRI, stress test,	
					pericardiocentesis,	
					myocardial biopsies, and	
					coronary angiography.	
					Continued on next page	

d frc ÷:+ Table A 1

.

	cy0bijectives and	Outcome Measures			Proposed a wearable	IoMT ECG, AI-driven	telehealth model for	arrhythmias of CVD	detection, prediction,	and management. The	proposed wearable ECG	can capture the	biosignal data and	analyse them using AI	and ML approaches.							Continued on next page	
ous page	Accura				99.56%.																		
inued from previ	Methods				Artificial	Neural	Network	(ANN),	Convolutional	Neural	Network	(CNN),	Support	Vector	Machine	(SVM),	K-Nearest	Neighbour	(KNN) and	Random	Forest (RF).		
Table A.1 $-$ cont	y Study	$\operatorname{Proposal}$			Design of a	secured	telehealth	system based	on multiple	biosignals	diagnosis and	classification.											
	Countr	of	Stud-	ies	France																		
	ID Year Author				$15 \ 2021 \ [21]$																		

zious page	Accuracy@bijectives and	Outcome Measures			93% for Designed a wearable	heart IoT phonogram (PCG),	sounds, an AI-driven telehealth	and model for the automatic	82% for estimation of heart rate	lung and breathing rate of	sounds. CVD patients.	Analysed a total of 88	10-second-long chest	sound samples from 76	preterm and full-term	babies.	Continued on next news
inued from prev	Methods				Support	Vector	Machine	(SVM),	K-Nearest	Neighbours	(KNN),	Decision Tree	(Tree).				
Table A.1 – conti	Country Study	of Proposal	Stud-	ies	Australia Neonatal heart	& lung sound	quality	assessment for	heart and	breathing rate	estimation.						
	ID Year Author				$16 \ 2020 \ [41]$												

ious page	Accuracy Wijectives and	Outcome Measures			UnspecifieRroposed a dynamic	ECG compatible with	telemedicine to prevent	and diagnose CVD	patients. The authors	applied ECG signal	analysing algorithms for	external noise reduction	captured by ECG.	Continued on next page	
nued from previ	Methods				Deep factor	analysis for	noise	reduction.							0
Table A.1 $-$ conti	Country Study	of Proposal	Stud-	ies	SingaporeECG signal	denoising	based on deep	factor analysis.							
	ID Year Author				$17 \ 2020 \ [44]$										

ous page	Accuracy (Ki) ectives and	Outcome Measures		63-93% Presented an invasive	CVD diagnostic	measurement and	evaluation of aortic	stiffness in the	carotid-femoral pulse	wave velocity (PWV)	index to diagnose	patients with CVDs.	The index explains how	the velocity of arterial	pulse moving along the	vessel wall indicates and	predicts the possible	CVD events. Used	cutting-edge AI (i.e.,	ANN) technology to	explore a new CVD	characteristics/elements	pattern.	Continued on next nage
nued from previ	Methods			Artificial	neural	networks	(ANNs).																	
Table A.1 $-$ conti	Country Study	of Proposal	Stud- ies	France Coronary	heart disease	diagnosis by	artificial neural	networks	including	aortic pulse	wave velocity	index and	clinical	parameters.										
	ID Year Author			18 2019 [45]	1																			

nous page	Accuracy@%jbctives and	Outcome Measures			UnspecifieRound that the CVD	risk is consistently	higher in indigenous	than non-indigenous	populations including	Native American,	Canadian First Nation,	Māori, Australian	Aboriginal, and Torres	Strait Islander	communities.	Continued on next page
nued from prev	Methods				Risk	assessment	and risk	management	algorithm.							
Table A.I – conti	Country Study	of Proposal	Stud-	ies	Australia Absolute CVD	risk and	lipid-lowering	therapy among	Aboriginal and	Torres Strait	Islander	Australians.				
	ID Year Author				19 2018 [7]											

• ر ا ۲ ۷ T.L.

ous page	Accuracy@bjbctives and Outcome Measures	UnspecifieRroposed a portable ECG-based telemedicine model for real-time monitoring of patients with CVD residing in non-clinical environments. The lightweight, portable ECG sensor enables sensing patients' heartbeat, amplitude level, and PQRST wave via a microcontroller Continued on next page
nued from previ	Methods	MultisimTM
Table A.1 $-$ contin	Country Study of Proposal Stud-	Pakistan Design and im- plementation of real-time electrocardio- gram monitoring system for telemedicine services.
	ID Year Author	20 2019 [42]

		Table A.1 – conti	nued from previo	is page
ID Year Author	Countr	y Study	Methods	Accuracy Wijectives and
	of	$\operatorname{Proposal}$		Outcome Measures
	Stud-			
	les			
$21 \ 2017 \ [55]$	Germany	y BlooXY: a	Used secured	UnspecifieDemonstrated an
		non-invasive	128 bits	efficient wearable sensor
		blood monitor.	a symmetric	called BlooXY for CVD
			encryption for	control, prevention,
			data security.	treatment, and
				management. BlooXY is
				able to sense and
				monitor blood pressure,
				heart rate, and blood
				oxygen level.
$22 \ 2019 \ [22]$	India	Plug and play	A self-	Unspecified ssessed hyper
		self-	configuration	connection with an
		configurable	algorithm is	ECG telemonitoring via
		IoT gateway	designed to	WBAN. Investigated
		node for	run at the	the adaptability of
		telemonitoring	gateway node	smartphones as an IoT
		of ECG.	to optimize the	gateway for sending and
			power	receiving data to a
			efficiency.	remote server.
				Continued on next page

• d frc +:+ Table A.1

rious page	Accuracy@%jectives and Outcome Measures		Overview Provided an overview of how AI platforms promote the operational delivery of care for patients with CVDs. Outlined strategies for integrating ML to augment existing clinical care processes to improve CVD care. Continued on next page
inued from prev	Methods		Overview
Table A.1 – conti	Country Study of Proposal Stud-	ies	USA Evaluation of AI and ML applications in real-world Cardiac Care.
	ID Year Author		23 2018 [56]

is page	Accuracy@bjbctives and Outcome Measures	UnspecifieBrovide a historical perspective of AI and ML approaches and evolution of deep learning algorithms through biomimicry (biological entities and processes). Main aim is to demystify these technology innovations for surgeons.	 98.3% Proposed a wearable ECG, AI-driven ECG, AI-driven telemedicine for AF detection and achieved 99.3% sensitivity, 97.4% specificity and 98.3% accuracy outperforming previous models. Continued on next page
nued from previou	Methods	Artificial neural deep learning.	Artificial neural network (ANN).
Table A.1 $-$ conti	Country Study of Proposal Stud- es	JSA To understand the impact of AI, ML, deep learning and cognitive computing on healthcare.	China Automated classification of atrial fibrillation.
	ID Year Author 6	24 2018 [57] U	25 2020 [60] (

rious page	Accuracy (kj) betives and	Uutcome Measures	UnspecifieRroposed a wearable	IoT-driven ECG for	CVD monitoring	systems. ECG data are	gathered using a	wearable monitoring	node and are	transmitted directly to	the IoT cloud using	WI-Fi. HTTP and	MQTT protocols are	employed to provide	visual and timely ECG	data.	Continued on next page	
nued from prev	Methods		ECG	monitoring	system.													
Table A.1 – conti	intry Study	Proposal d-	aa An IoT-cloud	based wearable	ECG	monitoring	system.											
	r Cou	of Stue ies	Chir															
	Autho		[48]															
	Year		2016															
	Ð		28															

vious page	Accuracy@bjbctives and	Outcome Measures			Overview Reviews the latest	developments in	surveillance technologies	for CVD. A variety of	signals that can monitor	CVD are summarised.	The different	mechanisms and	principles of monitoring	pulse signals are	discussed.	Continued on next page
Table A.1 – continued from previ	Methods				Overview											
	Country Study	of Proposal	Stud-	ies	China Wearable	sensors and	devices for	real-time CVD	monitoring.							
	ID Year Author				29 2021 [35]											

vious page	Accuracy@%jectives and	Outcome Measures			90.5% Designed wearable, wireless-based ECG watch (writs-worn) for the real-time monitoring. The proposed algorithm detects possible atrial fibrillation episodes within 10 seconds through a smartphone or desktop App.
nued from pre	Methods				Atrial fibrillation algorithm.
Table A.1 – conti	Country Study	of Proposal	Stud-	ies	Italy ECG WATCH: a real-time wireless wearable ECG.
	ID Year Author				30 2019 [43]

Appendix B. Characteristics of the Selected Studies.
1 2018 [49] Poland CVD BiocNemical120 N Y Smart health 2 2019 [50] Iran CVD BiocNemical120 N Y Y 3 2019 [51] USA CVD FourY 0 N Y Y 4 2020 [52] India CVD PonNcardiog000h Y Y Y 5 2020 [52] India CVD PonNcardiog000h Y Y Y 6 Sig- nal N Y Y Y Y 5 2020 [52] India CVD PonNcardiog000h Y Y Y 6 Sig- nal Sig- N Y Y Y 5 ECG N Y Y Y Y Y Y 5 ECG N Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y		U I EAI	Authors	Count	rlyisease Type	Devicés/See used(Y/N)	Size	eML/AI (Y/N)	Algoritl (Y/N)	uffielehealth/ Telemedicine/ <u>e-</u>
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$										bealth/ Smart health
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		2018	[49]	Poland	CVD	Bioclivemical shirt- based ECG	120	Z	Y	Y
 3 2019 [51] USA CVD WearMble 0 N Y 4 2020 [52] India CVD PhonMcardio3060 Y Y Y sig- ball ECG 	5	2019	[50]	Iran	CVD	FourY lay- ers ECG	0	Z	Y	Y
4 2020 [52] India CVD PhonMcardio 3060 Y Y Y sig- nal ECG	က	2019	[51]	USA	CVD	WearMble ECG		Z	Z	Υ
	4	2020	[52]	India	CVD	PhonMcardio sig- nal ECG	30000	X	А	Y

Table B.2: Characteristics of the Selected Studies.

	h/ cine/					Continued on next page
	hffielehealt Telemedi î	e- health/ Smart health	Z	X	Y	
ious page	$\frac{Algorit}{(Y/N)}$		Y	×	Ą	
from prev	IAL/AI (Y/N)		Y	z	Y	
continued	8/ Bethar in /N)Size		579	576	400	
ble B.2 –	Dev ic d used(Y		The N Car- diac Score- card sys-	tems Mob N é Tablet based CVD man- age- ment	IoT Y based ECG	
Ta	ttriðisease Type		CVD	alâVD	CVD	
	Cour	5	USA	Austr	India	
	Authors		[53]	[16]	[15]	
) Year		2016	2017	2021	
	I		ъ	9	2	

	fielehealth/ Telemedicine/	e- health/	Smart health	Y				Y							Continued on next page		
page	gorithi '/N) '	•		,				r									
evious	IV IV			Υ				У									
from pre	IdML/A (Y/N)			Y				Υ									
ontinued	' Forkan ip N) Size			ъ				317									
le $B.2 - cc$	DevicéB/ used(Y/I			IoT Y	based wrist-	worn FCC	sen-	sor IoT Y	based	ECG	Smart	Vest	sys-	tem			
Tab	ij) isease Type			CVD				CVD									
	Countr			UK				China									
	ar Authors			[8 [12]				[9 [91]	۲ ۲								
	[D Ye			3 201) 201									
				$ \infty $				0									

	ffielehealth/ Telemedicine/	ф-	health/	Smart	health	Υ								Υ			Υ				Continued on next page	
ous page	Algorith (Y/N)					Ν								Ν			Υ					
from previ	(Y/N)					N								N			Υ					
le B.2 - continued :	Devices/Secting used(Y/N)Size					CardNvascular	im-	plantable	elec-	tronic	de-	vices	(CIED)	IoT Y 0	sen-	SOIS	Phonk cardio & am	sig-	nal	ECG		
Tab	Countribisease Type					Poland CVD								MalaysiCVD			India CVD					
	ID Year Authors					$10 \ 2016 \ [13]$								$11 \ 2021 \ [19]$			$12 \ \ 2020 \ \ [14]$					

	elehealth/	elemedicine/	- 141- /	salth/	mart	ealth																	Continued on next page	
e	ithm) 	בל	ň	ກັ,	he	Ζ					Υ			Z									
ious page	Algori	(\mathbf{Y}/\mathbf{N})					Υ					Υ			Y									
from previ	IdML/AI	(\mathbf{Y}/\mathbf{N})					Ν					Y			Ν									
inued	Stature 2.	Size					0					129			12									
ole B.2 – cont	Devices/Se	used Y/N)					IoT N	aided	ECG	sen-	sor	DynaMic	ECG	Sor	WearMble	point-	ot-	care	imag-	ing	sys-	tem		
Tab	ıty isease	Type					CVD					rCVD			CVD									
	Count						China					Belgiur			USA									
	Authors						[48]					[47]			[46]									
	Year						2016					2020			2019									
	ID						13					14			15									

fielehealth/ Telemedicine/	e- health/ Smart health M	7	r Continued on next page	
uus page Algorith (Y/N)		-		
om previc eML/AI (Y/N)	>		X	
ontinued fn Boßtängel N) Size	010		4(3	
le B.2 – cc Devlcás/ used(Y/l	ж С	brease ar- te- te- cal- cal- fi- fi- mo- mo- mo- sram sges	based ECG	
Tab untı iy isease Type			nce UVD	
LS CO			в Н	
ID Year Autho	16 0017 [61]		1/ 2021 [41]	

												l on next page		
	uffelehealth/ Telemedicine/	e-	health/	Smart health	Υ		N		Ν	N		Continued		
ous page	$\begin{array}{c} Algorith \\ (Y/N) \end{array}$				Υ		Y		Ν	N				
from previ	(Y/N)				Υ		Υ		Ν	N				
ntinued	Eetkán ip V)Size				47		437		14350	0	0			
ole B.2 – co	Devices/ used(Y/N				ECGN	Sen- sor	DynaMnic ECG	Sen-	sor Not N	used Not N	used			
Tab	Countr i yisease Type				SingapoceVD		France CVD		AustraliaVD	AustraliaVD				
	ur Authors				0 [44]		0 [45]		8 [7]	5 [17]	-			
	ID Ye				18 202		19 202		$20 \ 201$	$21 \ 201$				

78

						on next page
	felehealth/ Telemedicine/ >-	nealth/ Smart	aealth Y		7.	Continued
ous page	$\frac{\text{Algorith}f}{(Y/N)}$		T A			
from previc	lteML/AI (Y/N)		Ν		z	
e B.2 – continued	Dev IcE/EoStan used(Y/N)Size		Atméğa- 0	32 ni- zro con- con- based hand- neld cCG	loT Y 37 Dased Writs- worn ECG (BlooXY)	
Tabl	Countribisease] Type 1		Pakista@VD		German&VD	
	ID Year Authors		22 2019 [42]		23 2017 [55]	

							ued on next page
	.hrffielehealth/ Telemedicine/ e-	health/ Smart health	Y	Z	Ν	N	Contin
ous page	Algorit (Y/N)		Y	¥	Y	¥	0
rom previ	léML/AI (Y/N)		Y	А	z	A	
ntinued f	Bolkan ia V) Size		549	-	0	0	
le B.2 – co	Devices/		IoT Y based ECG Sen- sor	IoT Y based ECG (wrist- worn)	WearMble ECG	WearMble ECG	
Tab	tyisease Type		CVD	CVD	CVD	CVD	
	Counti		India	USA	USA	China	
	ithors			_		_	
	ear Aı		19 [22	19 [56	18 [57	120 [60	
	IDY		24 20	25 20	$26 \ 20$	$27 \ 20$	

ous page	Algorithffielehealth/ (Y/N) Telemedicine/	¢-	health/	Smart	health	Y N						Y N			Y Y							Continued on next page	
n previo	(IL/AI (/N)																						
l fror	Vale ()					Υ						Х			Υ								
ontinued	/Bestann N) Size					0						0			0				G/ACG				
le $B.2 - c$	Devicés used(Y/					IoT Y	\mathbf{based}	HHMSP	sys-	tems	(ECG)	ECGN	sen-	SOT	IoT Y	\mathbf{based}	ECG,	PCG	SCG/BC	sen-	SOTS		
Tab	riyisease Type					CVD						CVD			CVD								
	Count					USA						USA			China								
	r Authors) [58]						§ [54]			[35]								
	Yea					2020						2018			2021								
	ID					28						29			30								

81

Table B.2 - continued from previous page Year Authors Countribisease Devices/BeakingleML/AI Algorithmelehealth/	Type used(Y/N)Size (Y/N) (Y/N) Telemedicine/ e- Smart health	2016 [48] China CVD Wearkble 0 Y Y Y ECG Sen- sors	2019 [43] Italy CVD IoT Y 40 Y Y Y ECG Watch	e: CVD=cardiovascular diseases; ECG=Electrocardiogram; PCG=Phonocardiogram; IoT= Internet of Things; T=Internet of Medical Things; ML=Machine Learning; AI=Artificial Intelligence	
ID Yea		31 201	32 201	Note: IoMT=	

82

Declaration of interests

☑ The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

□ The authors declare the following financial interests/personal relationships which may be considered as potential competing interests: